BI-ALGEBRAIC GEOMETRY AND THE ANDRÉ-OORT CONJECTURE

B. KLINGLER, E.ULLMO, A.YAFAEV

Contents

4 MI A 170 C C C C C C C C C C C C C C C C C C C	_
1. The André-Oort conjecture	2
1.1. The Hodge theoretic motivation	2
1.2. The André-Oort conjecture for \mathbb{C}^2	3
1.3. The Conjecture	4
1.4. History and results	4
1.5. Organization of the text	6
2. Pure Shimura varieties and their special subvarieties	8
3. Special structures on algebraic varieties	10
3.1. Pre-special structure	10
3.2. Special structure	10
3.3. Weakly special subvarieties	11
4. Bi-algebraic geometry and the Ax-Lindemann property.	12
4.1. Complex bi-algebraic geometry	12
4.2. Q-bi-algebraic geometry	14
4.3. Special structures and bi-algebraic structures	16
4.4. The Ax-Lindemann conjecture	16
5. O-minimal geometry and the Pila-Wilkie's theorem	17
5.1. O-minimal structures	17
5.2. Pila-Wilkie's counting theorem	18
6. O-minimality and Shimura varieties	20
6.1. Definability of π restricted to a fundamental set	20
6.2. Heights of special points	21
7. Strategy of the proof of the André-Oort conjecture for A_q	22
7.1. Proof of Theorem 7.1	23
7.2. Proof of Theorem 7.2	23
8. The hyperbolic Ax-Lindemann conjecture	25
8.1. Equivalence of (i) and (ii)	25
8.2. Stabilizers of maximal algebraic subvarieties of $\pi^{-1}(W)$.	26
8.3. O-minimal arguments and hyperbolic geometry	27
8.4. An algebraic curve of X^+ meets many fundamental domains	28
9. Lower bounds for Galois orbits of CM-points	29
9.1. Class groups for tori and reciprocity morphisms	29
9.2. Faltings height	30
9.3. Lower bounds for Galois orbits	31

9.4.	Colmez conjecture	32
10.	Further developments: the André-Pink conjecture.	34
References		35

1. The André-Oort conjecture

The André-Oort conjecture describes the distribution of special points on Shimura varieties. It is the analog in a Hodge-theoretic context of the Manin-Mumford conjecture (a theorem of Raynaud [Ray88]) stating that an irreducible subvariety of a complex Abelian variety containing a Zariski-dense set of torsion points is the translate of an Abelian subvariety by a torsion point. This text is a survey of the André-Oort conjecture, its context and its proof in the case of the Shimura variety \mathcal{A}_g moduli space of principally polarized complex Abelian varieties of dimension g (and more generally for mixed Shimura varieties whose pure part is of Abelian type) following a strategy proposed by Pila and Zannier and obtained through the work of many authors.

1.1. The Hodge theoretic motivation. Let us start by explaining the algebrogeometric problem underlying the conjecture. Let $f: \mathcal{X} \longrightarrow S$ be a smooth family of algebraic varieties over a quasi-projective smooth base S. Can we describe the locus of points $s \in S$ where the fiber \mathcal{X}_s (and its Cartesian powers) contain more algebraic cycles than the very general fiber (and its Cartesian powers)? We work over \mathbb{C} and consider the Hodge incarnation of this problem. Let $\mathbb{V} \to S$ be an admissible variation of mixed \mathbb{Z} -Hodge structures on the complex quasi-projective smooth base S (cf. [PS08, Def. 14.49]). In particular \mathbb{V} is a \mathbb{Z} -local system on S such that each fiber \mathbb{V}_s , $s \in S$, carries a graded-polarized mixed Hodge structure. This is an abstraction of the geometric case corresponding to $\mathbb{V} = (R^p f_* \mathbb{Z})_{\text{prim}}$ (for some p > 0) for f as above. One wants to understand the Hodge locus $\text{HL}(S, \mathbb{V}) \subset S$, namely the subset of points s in S for which exceptional Hodge classes of type (0,0) do occur in some $\mathbb{V}_{\mathbb{Q},s}^a \otimes (\mathbb{V}_{\mathbb{Q},s}^\vee)^b$, where $\mathbb{V}_{\mathbb{Q},s}^\vee$ denotes the \mathbb{Q} -Hodge structure dual to $\mathbb{V}_{\mathbb{Q},s}$.

The Tannakian formalism available for Hodge structures is particularly useful for describing $\operatorname{HL}(S,\mathbb{V})$. Recall that for every $s\in S$, the Mumford-Tate group $\operatorname{\mathbf{MT}}_s$ of the Hodge structure $\mathbb{V}_{\mathbb{Q},s}$ is the Tannakian group of the subcategory $<\mathbb{V}_{\mathbb{Q},s}^{\otimes}>$ of the Tannakian category of pure Hodge structures tensorially generated by $\mathbb{V}_{\mathbb{Q},s}$ and $\mathbb{V}_{\mathbb{Q},s}^{\vee}$. Equivalently, the group $\operatorname{\mathbf{MT}}_s$ is the stabiliser of the Hodge classes of type (0,0) in the rational Hodge structures tensorially generated by $\mathbb{V}_{\mathbb{Q},s}$ and its dual. A point $s\in S$ is said to be Hodge generic if $\operatorname{\mathbf{MT}}_s$ is maximal when s varies in its connected component. If S is connected, two Hodge generic points of S have the same Mumford-Tate group, called the generic Mumford-Tate group $\operatorname{\mathbf{MT}}_{S,\mathrm{gen}}$ of (S,\mathbb{V}) . The Hodge locus $\operatorname{HL}(S,\mathbb{V})$ is the subset of points of S which are not Hodge generic.

A fundamental result of Cattani-Deligne-Kaplan [CDK95] states that $\operatorname{HL}(S, \mathbb{V})$ is a countable union of closed irreducible algebraic subvarieties of S, each not contained in the union of the others. The irreducible components of the intersections of these algebraic subvarieties are called *special subvarieties* of (S, \mathbb{V}) . Hodge subvarieties of dimension zero are called *special points* of (S, \mathbb{V}) . We would like to understand the distribution of special points in S.

1.2. The André-Oort conjecture for \mathbb{C}^2 . The André-Oort conjecture answers this question when S is a Shimura variety. We start with its most explicit incarnation.

The simplest Shimura variety is the classical modular curve Y(1). As a complex analytic space it is the quotient $Y(1) := \mathbf{SL}_2(\mathbb{Z}) \setminus \mathcal{H}$, where $\mathcal{H} = \{ \tau \in \mathbb{C} : \operatorname{Im}(\tau) > 0 \}$ is the Poincaré upper-half plane and the group $\mathbf{SL}_2(\mathbb{Z})$ acts on \mathcal{H} by:

$$\left(\begin{smallmatrix} a & b \\ c & d \end{smallmatrix}\right)\tau = \frac{a\tau + b}{c\tau + d}.$$

The space Y(1) can also be interpreted as the set of complex elliptic curves up to isomorphism:

$$\operatorname{SL}_2(\mathbb{Z}) \backslash \mathcal{H} \longrightarrow \{E/\mathbb{C}\}/\cong , \qquad \tau \mapsto [E_\tau := \mathbb{C}/(\mathbb{Z}\tau + \mathbb{Z})] .$$

As complex elliptic curves up to isomorphism are classified by their j-invariant, the quotient map $\pi: \mathcal{H} \longrightarrow Y(1)$ identifies with the holomorphic j-map $j: \mathcal{H} \longrightarrow \mathbb{C}$ given by

$$\tau \mapsto j(E_{\tau}) = q^{-1} + 744 + 196884q + \cdots, \quad q = e^{2\pi i \tau}.$$

Hence the quotient $Y(1) \stackrel{\mathcal{I}}{\simeq} \mathbb{C}$ is the moduli space of complex elliptic curves; it is an algebraic variety, defined over \mathbb{Q} (even over \mathbb{Z}) because we have the notion of elliptic curves over arbitrary schemes.

Ignoring the stacky issues, the universal family of elliptic curves over $Y(1) \simeq \mathbb{C}$ defines a Hodge locus in Y(1), i.e. special points. For $\tau \in \mathcal{H}$, $\operatorname{End}(E_{\tau}) = \{z \in \mathbb{C} : z \cdot (\mathbb{Z}\tau + \mathbb{Z}) \subset \mathbb{Z}\tau + \mathbb{Z}\}$. Hence $\operatorname{End}(E_{\tau}) = \mathbb{Z}$ if $\dim_{\mathbb{Q}} \mathbb{Q}(\tau) \neq 2$ and $\operatorname{End}(E_{\tau})$ is an order in $\mathbb{Q}(\tau)$ if $\dim_{\mathbb{Q}} \mathbb{Q}(\tau) = 2$, in which case E_{τ} is a CM-elliptic curve. It follows easily that the Mumford-Tate group at $j(\tau)$ is $\operatorname{GL}(2,\mathbb{Q})$ in the first case, while it is $\operatorname{Res}_{\mathbb{Q}(\tau)/\mathbb{Q}} \operatorname{G}_m$ in the second. Hence special points (also called CM-points) in \mathbb{C} correspond to imaginary quadratic τ 's in \mathcal{H} , in particular they are dense (even for the analytic topology) in \mathbb{C} .

Let us now consider $Y(1)^2 \simeq \mathbb{C}^2$ as the moduli space of pairs of elliptic curves. Once more the Hodge locus for this family can be explicitly described:

- a point $x = (x_1, x_2) \in \mathbb{C}^2$ is special if both $x_1 \in \mathbb{C}$ and $x_2 \in \mathbb{C}$ are special.
- a special curve is either a line $\{x_1\} \times \mathbb{C}$ with x_1 special, a line $\mathbb{C} \times \{x_2\}$ with x_2 special, or the image T_n in \mathbb{C}^2 of the modular curve Y(n) parametrizing isogenies $\mathbb{Z}/n\mathbb{Z} \hookrightarrow E_1 \twoheadrightarrow E_2$ between two elliptic curves. The curve T_n is obtained from Y(n) by forgetting the isogeny (an equivalent definition of T_n is given below).

Each of these special curves contains a dense set of special points. Conversely André [An89] conjectured:

Conjecture 1.1. Let $\Sigma \subset \mathbb{C}^2$ be a set of special points, and let Z be an irreducible component of the its Zariski-closure $\overline{\Sigma}^{Zar}$. Then Z is one of the following:

- (1) a special point,
- (2) $\{x_1\} \times \mathbb{C}$ with x_1 special,
- (3) $\mathbb{C} \times \{x_2\}$ with x_2 special,
- (4) the image T_n (a Hecke correspondence) of

$$t_n \colon \mathcal{H} \to \mathcal{H} \times \mathcal{H} \to \mathbb{C}^2, \quad \tau \mapsto (\tau, n\tau) \mapsto (j(\tau), j(n\tau))$$

for some $n \in \mathbb{Z}_{>1}$,

(5) \mathbb{C}^2 itself.

Conjecture 1.1 was proven by Edixhoven [Ed98] under the Generalized Riemann Hypothesis (GRH) and by André [An98] unconditionally.

1.3. The Conjecture. We turn to the general case, first in an informal way. A pure Shimura variety S (resp. a mixed Shimura variety) is a complex quasi-projective moduli space of pure polarized (resp. mixed graded-polarized) Hodge structures with additional data, such that the universal family above S defines an admissible variation of (mixed) Hodge structure $\mathbb V$ over S. As explained by Deligne [De79] this restricts severely the possible types of Hodge structures we can consider. The prototype of a pure Shimura variety is the moduli space $\mathcal A_g$ of principally polarized Abelian varieties of dimension g, the variation $\mathbb V$ over $\mathcal A_g$ is the Hodge incarnation $R^1f_*\mathbb Z$ of the universal Abelian variety $f:\mathfrak A_g\longrightarrow \mathcal A_g$ over $\mathcal A_g$. The prototype of a mixed Shimura variety is $\mathfrak A_g$, the variation $\mathbb V$ over $\mathfrak A_g$ is the Hodge incarnation of the universal semi-Abelian variety over $\mathfrak A_g$.

As in Section 1.1 the variation \mathbb{V} over S defines special subvarieties in S. A special point of \mathcal{A}_g , also called a CM-point, corresponds to an Abelian variety with Complex Multiplication (CM). A special point on \mathfrak{A}_g is a torsion point on a CM-Abelian variety.

A crucial feature of Shimura varieties is their purely group-theoretic description: any Shimura variety S is defined thanks to a Shimura datum (\mathbf{G}, X) , where \mathbf{G} is a connected linear algebraic group over \mathbb{Q} and X is a certain homogeneous space under a subgroup of $\mathbf{G}(\mathbb{C})$. Special subvarieties of S also have a purely group theoretic description: they are precisely the images of the natural morphisms between Shimura varieties. In the next section we review this formalism for pure Shimura varieties.

Once we know that any Shimura variety S contains one special point, this group-theoretic formalism also implies that any special subvariety of S contains a dense (even for the Archimedian topology) set of special points. The André-Oort conjecture is the converse statement:

Conjecture 1.2 (André-Oort). Let Z be an irreducible subvariety of a mixed Shimura variety S. If Z contains a Zariski-dense set of special points then Z is a special subvariety of S.

1.4. **History and results.** Motivated by transcendence questions about periods of Shimura varieties, André [An89, p.215, Problem 1] formulated Conjecture 1.2 for a curve Z contained in a pure Shimura variety. Oort [Oort94] was interested in the study of Jacobians with complex multiplication and proposed Conjecture 1.2 for $S = A_g$. Hence the name of the conjecture.

Both André and Oort were aware of the analogy with the Manin-Mumford conjecture. This analogy has inspired all the strategies for proving Conjecture 1.2.

- (a) The p-adic methods of Raynaud's proof [Ray88] of the Manin-Mumford conjecture inspired works on Conjecture 1.2 when S is a pure Shimura variety and for the Zariski-closure Z of a set of special points having good reduction properties at one fixed place p [Moo98,II], [Ya05].
- (b) Edixhoven developed an approach to Conjecture 1.2, based on Galois techniques and intersection theory, retrospectively close in spirit to Hindry's approach to the Manin-Mumford conjecture [Hin88]. This method uses in a crucial way effective Cebotarev type results, known only under the Generalized Riemann Hypothesis (GRH). In [Ed98]

Edixhoven proves Conjecture 1.2 under GRH for S a product of two modular curves; in [EdYa03] Edixhoven and Yafaev obtain the result under GRH for Z a curve in an arbitrary pure Shimura variety S; and in [Ed05] Edixhoven proves Conjecture 1.2 under GRH for Z an arbitrary subvariety of a product of modular curves. This approach, allied with ideas à la Margulis-Ratner from ergodic theory on homogeneous spaces ([CloUl05], [U07]), culminated in the following result [UY14a], [KY14] (announced in 2006 and published in 2014):

Theorem 1.3. The André-Oort Conjecture 1.2 for pure Shimura varieties is true under the Generalized Riemann Hypothesis. It is also true unconditionally if Z is the Zariski-closure of a set of special points contained in a Hecke orbit.

This text will say nothing about Edixhoven's approach, for which many surveys are available. We refer in particular to [Ya07] and [Panorama] and the references therein.

(c) Pila and Zannier [PiZa08] developed a method based on o-minimal geometry for proving the Manin-Mumford conjecture. Pila adapted it to obtain an unconditional proof of Conjecture 1.2 for S an arbitrary product $\mathbb{C}^n \times \mathbf{G}_m^k$ [Pil11] (as we already mentioned, André obtained an unconditional proof for S the product of two modular curves but his method using Puiseux expansion did not generalize). The combination of the work of many authors (whose contributions are detailed below) then lead to the following:

Theorem 1.4. The André-Oort Conjecture 1.2 is true for A_g and more generally for any mixed Shimura variety whose pure part is of Abelian type.

The goal of this text is to present the ideas around Conjecture 1.2 and sketch the proof of Theorem 1.4 following the Pila-Zannier strategy. Following [U14], Conjecture 1.2 for a general connected mixed Shimura variety S uniformized by $\pi: X^+ \longrightarrow S := \Gamma \backslash X^+$ follows from three main ingredients (two of which are known in full generality while the third one is known only under GRH or unconditionally for mixed Shimura varieties whose pure part is of Abelian type):

The first ingredient is the definability in the o-minimal structure $\mathbb{R}_{\mathrm{an,exp}}$ of the restriction of π to a semi-algebraic fundamental set \mathcal{F} for the action of Γ on X^+ : see Theorem 6.2. This result is obtained by Peterzil-Starchenko [PetStar13] for $S = \mathcal{A}_g$, by Klingler-Ullmo-Yafaev [KUY16] for an arbitrary pure Shimura variety and extended by Gao [Gao16] to any mixed Shimura variety.

The second ingredient is the Ax-Lindemann conjecture for Shimura varieties, see Theorem 4.25, which says that the Zariski-closure $\pi(Y)$ of any algebraic subvariety Y of X^+ (in the sense of Example 4.6) should be weakly special (in the sense of Section 3.3). This is the main geometric ingredient in the Pila-Zannier strategy for solving the Manin-Mumford-André-Oort problem for Shimura varieties.

Theorem 4.25 is proven by Pila [Pil11] when S is a product $Y(1)^n \times (\mathbb{C}^*)^k$, by Ullmo-Yafaev [UY14b] for projective Shimura varieties, by Pila-Tsimerman [PT14] for \mathcal{A}_g , by Klingler-Ullmo-Yafaev [KUY16] for any pure Shimura variety and extended by Gao [Gao16] to any mixed Shimura variety. All these proofs use o-minimal geometry as a tool. Mok has an entirely complex-analytic approach to the Ax-Lindemann conjecture in the pure case. We refer to [Mok10], [Mok12] for partial results.

The third ingredient is a good lower bound for the size of Galois orbits of special points of S. This ingredient is already crucial in the Edixhoven's approach. We refer to Section 9 for the description of the expected lower bound for an arbitrary pure Shimura variety. These expected lower bounds are known under GRH for any pure Shimura variety following results of Tsimerman [Tsi12] and Ullmo-Yafaev [UY15]. They are known unconditionally only for mixed Shimura varieties whose pure part is of Abelian type. For simplicity we restrict ourselves to the case $S = \mathcal{A}_g$. Given a point $x \in \mathcal{A}_g$ let A_x be the principally polarized Abelian variety parametrized by x and d_x the absolute value of the discriminant of the center of the ring of endomorphisms of A_x . When x is special its field of definition k(x) is a number field. One wants to show that there exist real positive numbers $\alpha = \alpha(g)$ and $\beta = \beta(g)$ such that for any special point $x \in \mathcal{A}_g$ one has:

$$[k(x):\mathbb{Q}] > \alpha \cdot d_x^{\beta} .$$

Tsimerman [Tsi] remarkably noticed that the inequality (1.1) follows from the Masser-Wüstholz isogeny theorem [MaWü95] and an upper bound for the Faltings height $h_F(A_x)$ of the form

$$(1.2) \forall \epsilon > 0, \quad h_F(A_x) \ll_{\epsilon} d_x^{\epsilon} .$$

Colmez [Col93] conjectured a closed formula for the Faltings height of an Abelian variety with complex multiplication, depending only on its CM-type (E, Φ) . Fixing E and averaging on the 2^g possible CM-type Φ for E one obtains a simpler formula for the average of the Faltings height of Abelian varieties with CM by the ring of integers \mathcal{O}_E of E. The upper-bound (1.2) follows from this Colmez conjecture on average and classical arguments from analytic number theory (see [Tsi]).

Finally a proof of Colmez conjecture on average has been obtained independently by Andreatta-Goren-Howard-Madapusi Pera [AGHM] (studying CM-points on certain orthogonal Shimura varieties) and Yuan-Zhang [YuZh] (analyzing Heegner points on certain Shimura curves).

Daw-Orr [DawOrr15] show that the Pila-Zannier method gives a new proof of Conjecture 1.2 under GRH for an arbitrary pure Shimura variety.

Gao [Gao16] extends the Pila-Zannier method in the mixed setting, showing Conjecture 1.2 under GRH for any mixed Shimura variety and Conjecture 1.2 unconditionally for mixed Shimura varieties whose pure part is of Abelian type.

1.5. **Organization of the text.** This text is organized as follows.

For the reader's convenience Section 2 gives a short introduction to the formalism of Shimura varieties using Deligne's language of Hodge theory. For simplicity we restrict ourselves to the *pure Shimura varieties*.

Section 3 describes a general format where a reasonable Manin-Mumford-André-Oort type problem can be formulated: the notion of a *special structure* on a complex algebraic variety S, which axiomatizes the properties of the collection of special subvarieties on a Shimura variety or an Abelian variety. We also notice that in all the cases we consider, special structures are related to Kähler geometry through the notion of *weakly special subvarieties*: in the case of semi-abelian varieties or pure Shimura varieties, weakly special subvarieties are exactly the totally geodesic subvarieties for the canonical Kähler

metric on S. The special subvarieties of S are precisely the weakly special ones (a purely geometric notion) containing a smooth special point (an arithmetic notion).

This dichotomy between geometry and arithmetic persists in Section 4, where we develop the idea of bi-algebraic geometry. In a few words, this is the study of the transcendence properties of the uniformization morphism $\pi: \tilde{S} \longrightarrow S$ for S a connected algebraic variety whose universal cover \tilde{S} can be endowed with an algebraic structure (in the precise sense of Definition 4.1). The bi-algebraic structure is non-trivial when the uniformization map π is non-algebraic. In this situation, bi-algebraic subvarieties of S are defined by a functional transcendence constraint: they are the irreducible algebraic subvarieties of S images of algebraic subvarieties of S (in the sense of Definition 4.2). All the special structures we consider are of bi-algebraic origin (see Section 4.3), and bi-algebraic subvarieties and weakly special subvarieties coincide. Hence special subvarieties are exactly the bi-algebraic structure can be enriched over \mathbb{Q} (see Section 4.2) and the special points are exactly the arithmetic bi-algebraic points (see Definition 4.10).

The geometry of non-trivial bi-algebraic structures is governed by a natural conjecture in functional transcendence: given a connected algebraic variety S with a bi-algebraic structure, the Ax-Lindemann conjecture predicts that the Zariski-closure $\pi(Y)$ of any algebraic subvariety Y of \tilde{S} should be bi-algebraic. This is the main geometric ingredient in the Pila-Zannier strategy for solving the Manin-Mumford-André-Oort problem for special structures of bi-algebraic origin.

In Section 5 we turn to the techniques at our disposal for attacking the Ax-Lindemann conjecture and the Manin-Mumford-André-Oort problem in general. Let S be an algebraic variety endowed with a bi-algebraic structure. Whether or not this bi-algebraic structure underlies a special structure on S seems to depend on the existence of a common geometric framework for S and \tilde{S} , more flexible than (semi-) algebraic geometry as the map $\pi: \tilde{S} \longrightarrow S$ is far from algebraic, but topologically more constraining than analytic geometry in order to explain the special structure. Such a common framework is reminiscent of Grothendieck's idea of "tame topology" [Gro84, section 5], and is described in model theoretic language as o-minimal geometry. Section 5 presents a minimal recollection of o-minimal geometry, and state a deep diophantine criterion due to Pila and Wilkie for detecting (positive dimensional) semi-algebraic subsets of \mathbb{R}^n among subsets definable in an o-minimal structure: if such a subset contains polynomially many (with respect to the height) points of \mathbb{Q}^n then it contains a non-trivial positive dimensional semi-algebraic subset.

The last four sections turn to the proof of the André-Oort conjecture Theorem 1.4 for pure Shimura varieties of Abelian type. Section 6 details the first ingredient, namely the o-minimal geometry of pure Shimura varieties which culminates in Theorem 6.2. Section 7 describes the general structure of the proof. Section 8 sketch the proof of the second ingredient: the hyperbolic Ax-Lindemann Theorem 4.25, known for any Shimura variety. Section 9 considers its arithmetic part: the obtention of good lower bounds for the size of Galois orbits of special points, known only for Shimura varieties of Abelian type.

This text is largely inspired by the course on the André-Oort conjecture given by E. Ullmo at IHES in Spring 2016. For other surveys on the André-Oort conjecture following the Pila-Zannier method, we refer to [Daw16] for a more elementary introduction, to [Sca12] and [Sca16] for the description of the method in the geometrically easier case of $S = \mathbb{C}^n \times \mathbf{G}_m^k$ but with an expanded treatment of the o-minimal background.

2. Pure Shimura varieties and their special subvarieties

References for this section are [De71], [De79], [Mi05]. For mixed Shimura varieties, see [Pink89].

Recall that a pure \mathbb{Q} -Hodge structure on a \mathbb{Q} -vector space V is a linear decomposition $V_{\mathbb{C}} = \bigoplus_{p,q \in \mathbb{Z}} V^{p,q}$ such that $\overline{V^{p,q}} = V^{q,p}$. Equivalently it is a morphism of real algebraic groups $h \colon \mathbf{S} \longrightarrow \mathbf{GL}(V_{\mathbb{R}})$, where $\mathbf{S} = \mathrm{Res}_{\mathbb{C}/\mathbb{R}} \mathbf{G}_{\mathbf{m},\mathbb{C}}$ denotes the Deligne's torus (hence $\mathbf{S}(\mathbb{R}) = \mathbb{C}^*$). The Mumford-Tate group $\mathbf{MT}(h)$ we defined in Section 1.1 is equivalently the smallest algebraic \mathbb{Q} -subgroup \mathbf{H} of $\mathbf{GL}(V)$ such that h factors through $\mathbf{H}_{\mathbb{R}}$. It is a reductive group if V is assumed to be polarized.

A Shimura datum is a pair (\mathbf{G}, X) , with \mathbf{G} a linear connected reductive group over \mathbb{Q} and X a $\mathbf{G}(\mathbb{R})$ -conjugacy class of a morphism of real algebraic groups $h \in \mathrm{Hom}(\mathbf{S}, \mathbf{G}_{\mathbb{R}})$, satisfying the "Deligne's conditions" [De79, 1.1.13]:

- (D1) The Hodge structure on the Lie algebra \mathfrak{g} defined by Ad $\circ h$ has Hodge types (-1,1), (0,0) and (1,-1) only.
- (D2) The conjugation by h(i) defines a Cartan involution of the group of real points $\mathbf{G}^{\mathrm{ad}}(\mathbb{R})$ of the adjoint group \mathbf{G}^{ad} : the subgroup $\{g \in \mathbf{G}^{\mathrm{ad}}(\mathbb{C}), h(i)^{-1}\overline{g}h(i) = g\}$ of $\mathbf{G}^{\mathrm{ad}}(\mathbb{C})$ is compact.
- (D3) for every simple factor \mathbf{H} of \mathbf{G} , the composition of $h: \mathbf{S} \longrightarrow \mathbf{G}_{\mathbb{R}}$ with the projection $\mathbf{G}_{\mathbb{R}} \longrightarrow \mathbf{H}_{\mathbb{R}}$ is non-trivial.

These conditions imply, in particular, that the connected components of X are Hermitian symmetric domains. Any Hermitian symmetric domain can be obtained in this way. A morphism of Shimura data from (\mathbf{G}_1, X_1) to (\mathbf{G}_2, X_2) is a \mathbb{Q} -morphism $f: \mathbf{G}_1 \longrightarrow \mathbf{G}_2$ mapping X_1 to X_2 .

Definition 2.1. Let (\mathbf{G}, X) be a Shimura datum and K a compact open subgroup of $\mathbf{G}(\mathbb{A}_f)$ (where \mathbb{A}_f denotes the ring of finite adèles of \mathbb{Q}). The Shimura variety $\mathrm{Sh}_K(\mathbf{G}, X)$ is the complex analytic space $\mathbf{G}(\mathbb{Q}) \setminus (X \times \mathbf{G}(\mathbb{A}_f)/K)$, where $\mathbf{G}(\mathbb{Q})$ acts diagonally on $X \times \mathbf{G}(\mathbb{A}_f)/K$.

Proposition 2.2. Let $\mathbf{G}(\mathbb{R})_+$ be the stabilizer in $\mathbf{G}(\mathbb{R})$ of a connected component X^+ of X and $\mathbf{G}(\mathbb{Q})_+ := \mathbf{G}(\mathbb{R})_+ \cap \mathbf{G}(\mathbb{Q})$. The class group $\mathcal{C} := \mathbf{G}(\mathbb{Q})_+ \setminus \mathbf{G}(\mathbb{A}_f)/K$ is finite and one has the decomposition

(2.1)
$$\operatorname{Sh}_{K}(\mathbf{G}, X) = \coprod_{g \in \mathcal{C}} \Gamma_{g} \backslash X^{+} ,$$

where Γ_g denotes the congruence arithmetic lattice $gKg^{-1} \cap \mathbf{G}(\mathbb{Q})_+$ of $\mathbf{G}(\mathbb{R})_+$.

Each $\Gamma_g \backslash X^+$ has finite volume for the natural (up to a non-zero multiple scalar) $\mathbf{G}(\mathbb{R})_+$ -invariant measure on the Hermitian symmetric space X^+ . It follows from results of Baily and Borel [BB66] that each $\Gamma_g \backslash X^+$ has a natural structure of complex

quasi-projective variety, hence also $\operatorname{Sh}_K(G,X)$. Moreover the natural analytic morphism $\operatorname{Sh}_{K_1}(\mathbf{G}_1,X_1) \longrightarrow \operatorname{Sh}_{K_2}(\mathbf{G}_2,X_2)$ deduced from a morphism of Shimura data $f:(\mathbf{G}_1,X_1) \longrightarrow (\mathbf{G}_2,X_2)$ mapping a compact open subgroup $K_1 \subset \mathbf{G}_1(\mathbb{A}_f)$ into $K_2 \subset \mathbf{G}_2(\mathbb{A}_f)$ is naturally algebraic.

If Γ_g has no torsion then the algebraic variety $\Gamma_g \backslash X^+$ is smooth. Usually we work with a stronger notion of neat compact open subgroup $K \subset \mathbf{G}(\mathbb{A}_f)$, in which case $\mathrm{Sh}_K(\mathbf{G},X)$ is smooth.

The quotient $S = \Gamma_e \backslash X^+$ is called the connected Shimura variety associated to the Shimura datum (\mathbf{G}, X) and the compact open subgroup $K \subset \mathbf{G}(\mathbb{A}_f)$.

The projective limit $\operatorname{Sh}(\mathbf{G},X)_{\mathbb{C}} = \lim_K \operatorname{Sh}_K(\mathbf{G},X)_{\mathbb{C}}$ is a \mathbb{C} -scheme on which $\mathbf{G}(\mathbb{A}_{\mathrm{f}})$ acts continuously by multiplication on the right. The multiplication by $g \in \mathbf{G}(\mathbb{A}_{\mathrm{f}})$ on $\operatorname{Sh}(\mathbf{G},X)$ induces an algebraic correspondence T_g on $\operatorname{Sh}_K(G,X)$, called a Hecke correspondence.

Let $\rho: \mathbf{G} \longrightarrow \mathbf{GL}(V)$ be a rational representation of \mathbf{G} . Choose a \mathbb{Z} -structure $V_{\mathbb{Z}}$ on V such that $\rho(K) \subset \mathbf{GL}(V_{\widehat{\mathbb{Z}}})$. Every point $x \in X$ defines a polarized \mathbb{Z} -Hodge structure

$$\rho \circ x : \mathbf{S} \xrightarrow{x} \mathbf{G}_{\mathbb{R}} \xrightarrow{\rho} \mathbf{GL}(V_{\mathbb{R}})$$

on $V_{\mathbb{Z}}$. These $\rho \circ x$, $x \in X$, aggregate to form a polarized variation of \mathbb{Z} -Hodge structure \mathbb{V}_{ρ} on $\mathrm{Sh}_K(\mathbf{G},X)$. The collection of special subvarieties on $\mathrm{Sh}_K(\mathbf{G},X)$ associated with \mathbb{V}_{ρ} is shown to be independent of the choice of the faithful representation ρ and has a purely group-theoretic description: a subvariety $V \subset \mathrm{Sh}_K(\mathbf{G},X)_{\mathbb{C}}$ is special if and only if there is a Shimura datum $(\mathbf{H},X_{\mathbf{H}})$, a morphism of Shimura data $f:(\mathbf{H},X_{\mathbf{H}}) \longrightarrow (\mathbf{G},X)$ and an element $g \in \mathbf{G}(\mathbb{A}_f)$ such that V is an irreducible component of the image of the Hecke-Shimura morphism, also called a Hecke-correspondence:

$$\operatorname{Sh}(\mathbf{H}, X_{\mathbf{H}}) \xrightarrow{\operatorname{Sh}(f)} \operatorname{Sh}(\mathbf{G}, X) \xrightarrow{g} \operatorname{Sh}(\mathbf{G}, X) \longrightarrow \operatorname{Sh}_K(\mathbf{G}, X)$$
.

It can also be shown that the Shimura datum $(\mathbf{H}, X_{\mathbf{H}})$ can be chosen in such a way that $\mathbf{H} \subset \mathbf{G}$ is the generic Mumford-Tate group on $X_{\mathbf{H}}$. A *special* point is a special subvariety of dimension zero. One sees that a point $[x, gK] \in \operatorname{Sh}_K(\mathbf{G}, X)$ (where $x \in X$ and $g \in \mathbf{G}(\mathbb{A}_f)$) is *special* if and only if the group $\mathbf{MT}(x)$ is commutative (in which case $\mathbf{MT}(x)$ is a torus).

Given a special subvariety S of $\operatorname{Sh}_K(\mathbf{G}, X)$, the set of special points of $\operatorname{Sh}_K(\mathbf{G}, X)(\mathbb{C})$ contained in S is dense in V for the strong (and in particular for the Zariski) topology. Indeed, one shows that S contains a special point, say s. Let \mathbf{H} be a reductive group defining S and let $\mathbf{H}(\mathbb{R})^+$ denote the connected component of the identity in the real Lie group $\mathbf{H}(\mathbb{R})$. The fact that $\mathbf{H}(\mathbb{Q}) \cap \mathbf{H}(\mathbb{R})^+$ is dense in $\mathbf{H}(\mathbb{R})^+$ implies that the " $\mathbf{H}(\mathbb{Q}) \cap \mathbf{H}(\mathbb{R})^+$ -orbit" of s, which is contained in S, is dense in S. This "orbit" (sometimes referred to as the Hecke orbit of s) consists of special points.

For the precise definition of a Shimura datum (resp. a Shimura variety) of Abelian type we refer to [Mi05, section 8]. We only recall that if (\mathbf{G}, X^+) is a connected Shimura datum with \mathbf{G} simple then:

- if \mathbf{G}^{ad} is of type A, B or C then (\mathbf{G}, X^+) is of Abelian type.
- if \mathbf{G}^{ad} is of type E_6 or E_7 then (\mathbf{G}, X^+) is not of Abelian type.
- if \mathbf{G}^{ad} is of type D then (\mathbf{G}, X^+) may or may not be of Abelian type.

10

3. Special structures on algebraic varieties

- 3.1. **Pre-special structure.** In this section we introduce a general format in which a Manin-Mumford-André-Oort type problem can be formulated: the notion of a special structure on an algebraic variety. We refer to [U16] for more details and [Zil13] for a study of special subvarieties from the point of view of model theory.
- **Definition 3.1.** (pre-special structure) Let S be a complex quasi-projective variety. A pre-special structure on S is the datum of a countable set $\Sigma(S)$ of irreducible algebraic subvarieties of S, called special subvarieties of S, satisfying the following properties:
 - (i) $S \in \Sigma(S)$, i.e. S is special.
 - (ii) an irreducible component of an intersection of special subvarieties of S is a special subvariety of S.
 - (iii) Let $\Sigma_i(S) \subset \Sigma(S)$ be the set of special subvarieties of S of dimension i. For any $W \in \Sigma(S)$, special points of S (i.e. elements of $\Sigma_0(S)$) are dense in W.

It follows from property (ii) that for any irreducible algebraic subvariety Z of S, there exists a unique smallest special subvariety of S containing Z. One says that Z is Hodge generic if it not contained in any strict special subvariety of S.

The following are natural examples of complex algebraic varieties endowed with a pre-special structure:

- (1) a complex semi-Abelian variety W extension of an Abelian variety A by a torus $T \simeq \mathbf{G}_m^n$, with special subvarieties the translate by a torsion point of an algebraic subgroup. Special points are torsion points.
 - (2) a Shimura variety S with its special subvarieties.
- 3.2. **Special structure.** Notice that many other examples of pre-special structures are obtained as follows. Suppose S is a quasi-projective variety over $\overline{\mathbb{Q}}$ and define an irreducible subvariety of $S_{\mathbb{C}}$ to be special if it is defined over $\overline{\mathbb{Q}}$. Such a pre-special structure is called trivial as there seems to be nothing interesting to say in general about the distribution of special points. We exclude such trivial examples by strengthening the definition of a pre-special structure as follows:

Definition 3.2. (special structure) Let S be a complex quasi-projective variety. A special structure on S is the datum of a pre-special structure on each S^r , $r \in \mathbb{Z}_{\geq 1}$, such that for any $W \in \Sigma(S)$ then W^r belongs to $\Sigma(S^r)$ and satisfies:

- (a) $\Sigma_0(W^r) = \Sigma_0(W)^r$.
- (b) for any r-tuple (a_1, \dots, a_r) then

$$\prod_{i=1}^r \Sigma_{a_i}(W) \subset \Sigma_a(W^r) ,$$

where $a = \sum_{i=1}^{r} a_i$.

(c) The set $\Sigma_{d_W}(W^2)$ contains a countable infinite set of bi-étale algebraic correspondences of W, where d_W denotes the dimension of W. Here an algebraic correspondence of W is understood to be an algebraic subvariety V of W^2 whose projections on both factors are finite and surjective; it is said to be bi-étale if moreover these projections are étale.

As any finite product of semi-Abelian varieties is a semi-Abelian variety and a finite product of Shimura varieties is a Shimura variety, one easily checks that the pre-special structure on a semi-Abelian variety or a Shimura variety is special. The only non-trivial condition is (c): the existence of infinitely many bi-étale algebraic correspondences is provided by endomorphisms of a semi-abelian variety, and Hecke correspondences of a Shimura variety.

An abstract Manin-Mumford-André-Oort type conjecture in the format defined above is one of the following equivalent statements (1) or (2) below (the equivalence follows from the properties (ii) and (iii) of pre-special varieties):

Conjecture 3.3. (Abstract Manin-Mumford-André-Oort) Let S be a complex quasiprojective variety endowed with a special structure.

- (1) Let Z be an irreducible algebraic subvariety of S containing a Zariski-dense set of special points. Then Z is a special subvariety of S.
- (2) Let Z be an algebraic subvariety of S. The set of special subvarieties of S contained in Z and maximal for these properties is finite.

Conjecture 3.3 in the case of semi-abelian varieties is the classical Manin-Mumford conjecture, while we recover the André-Oort Conjecture 1.2 in the case of Shimura varieties.

Remark 3.4. Notice that any semi-Abelian variety can be realized as a subvariety of a mixed Shimura variety. However only the ones whose Abelian part is CM can be realized a special subvarieties of a mixed Shimura variety. Hence the André-Oort Conjecture 1.2 implies the Manin-Mumford conjecture only for such semi-Abelian varieties. In [Zil02] and [Pink05], Zilber and Pink propose a general conjecture (now called the Zilber-Pink conjecture) about atypical intersections in mixed Shimura varieties, which implies both the Manin-Mumford and the André-Oort conjecture. We refer the reader to the volume [Panorama] for an exposition of the Zilber-Pink conjecture.

3.3. Weakly special subvarieties. This section relates special structures and Kähler geometry.

Notice first that any semi-Abelian variety A is endowed with an essentially canonical Kähler metric coming from the flat Euclidean metric on its uniformization \mathbb{C}^n . Define a weakly special subvariety of A as an irreducible algebraic subvariety whose smooth locus is totally geodesic in A. Thus special subvarieties are weakly special, and a weakly special subvariety is special if and only if it contains a special point.

Similarly, a connected pure Shimura variety S (assumed to be smooth) inherits an essentially canonical Kähler metric from its universal cover X^+ : any locally symmetric Kähler metric on the Hermitian symmetric space X^+ is invariant under Γ hence descends to $S = \Gamma \backslash X^+$. Notice that the locally symmetric Kähler metric on X^+ is unique (up to a scalar) if X^+ is irreducible as a symmetric space: it coincides with the Bergman metric of the bounded Harish-Chandra realization of X^+ .

Define once more a weakly special subvariety of S as an irreducible algebraic subvariety whose smooth locus is totally geodesic in S. Every special subvariety of S is easily seen to be weekly special. Similarly to the case of semi-Abelian varieties, Moonen [Moo98,I] proved:

Theorem 3.5. Let S be a pure connected Shimura variety. A weakly special subvariety of S is special if and only if it contains a special point.

More precisely: let $(\mathbf{H}, X_{\mathbf{H}})$ be a sub-Shimura datum of the Shimura datum (\mathbf{G}, X) defining S. Assume that the adjoint Shimura datum $(\mathbf{H}^{\mathrm{ad}}, X_{\mathbf{H}^{\mathrm{ad}}})$ splits as a product:

$$(\mathbf{H}^{\mathrm{ad}}, X_{\mathbf{H}^{\mathrm{ad}}}) = (\mathbf{H}_1, X_1) \times (\mathbf{H}_2, X_2)$$
.

Let x_2 be a point of X_2 and Z the image of $X_1^+ \times x_2$ in S. Then Z is weakly special, and Z is special if and only if x_2 is a special point of X_2 . Conversely any weakly special subvariety of S is obtained in this way.

When S is a general mixed Shimura variety, Pink [Pink05, def.4.1] defines the weakly special subvarieties of S in terms of mixed Shimura data. Once more the special subvarieties are exactly the weakly special ones containing one special point.

4. BI-ALGEBRAIC GEOMETRY AND THE AX-LINDEMANN PROPERTY.

4.1. Complex bi-algebraic geometry. Let X and S be (connected) complex algebraic varieties and suppose $\pi: X^{\mathrm{an}} \longrightarrow S^{\mathrm{an}}$ is a complex analytic, non-algebraic, morphism between the associated complex analytic spaces. In this situation the image $\pi(Y)$ of a generic algebraic subvariety $Y \subset X$ is usually highly transcendental and the pairs $(Y \subset X, V \subset S)$ of irreducible algebraic subvarieties such that $\pi(Y) = V$ are rare and of particular geometric significance. We are especially interested in the case where X is the universal cover \tilde{S} of S. In this case however, the requirement that \tilde{S} is a complex algebraic variety is too restrictive for practical purposes. We relax it as follows:

Definition 4.1. A bi-algebraic structure on a connected complex algebraic variety S is a pair

$$(D: \tilde{S} \longrightarrow \hat{X}, \quad h: \pi_1(S) \longrightarrow \operatorname{Aut}(\hat{X}))$$

where \tilde{S} denotes the universal cover of S, \hat{X} is a complex algebraic variety, $\operatorname{Aut}(\hat{X})$ its group of algebraic automorphisms, $h: \pi_1(S) \longrightarrow \operatorname{Aut}(\hat{X})$ is a group morphism (called the holonomy representation) and D is a local biholomorphism (called the developing map).

Definition 4.2. Let S be a connected complex algebraic variety S endowed with a bialgebraic structure (D,h).

- (i) An irreducible analytic subvariety $Y \subset \tilde{S}$ is said to be an irreducible algebraic subvariety of \tilde{S} if D(Y) is open (for the analytic topology) in its Zariski-closure $D(Y)^{\text{Zar}} \subset \hat{X}$.
- (ii) An irreducible algebraic subvariety $Y \subset \tilde{S}$, resp. $W \subset S$, is said to be bialgebraic if $\pi(Y)$ is an algebraic subvariety of S, resp. any (equivalently one) analytic irreducible component of $\pi^{-1}(W)$ is an irreducible algebraic subvariety of \tilde{S} .

Example 4.3. (Tori)

The paradigm of a bi-algebraic structure is provided by the multi-exponential

$$\pi := (\exp(2\pi i \cdot), \dots, \exp(2\pi i \cdot)) : \mathbb{C}^n \longrightarrow (\mathbb{C}^*)^n$$
.

In this case $\tilde{S} = \hat{X} = \mathbb{C}^n$ and D is the identity morphism. One easily shows that an irreducible algebraic subvariety $Y \subset \mathbb{C}^n$ (resp. $W \subset (\mathbb{C}^*)^n$)) is bi-algebraic if and only

if Y is a translate of a rational linear subspace of $\mathbb{C}^n = \mathbb{Q}^n \otimes_{\mathbb{Q}} \mathbb{C}$ (resp. W is a translate of a subtorus of $(\mathbb{C}^*)^n$).

For the choice of the factor $2\pi i$ in the exponential, see Section 4.2.

Example 4.4. (Abelian varieties)

Let $\pi: \text{Lie}A \simeq \mathbb{C}^n \longrightarrow A$ be the uniformizing map of a complex Abelian variety A of dimension n. Once more $\tilde{S} = X = \mathbb{C}^n$ and D is the identity morphism. One checks that an irreducible algebraic subvariety $W \subset A$ is bi-algebraic if and only if W is the translate of an Abelian subvariety of A (cf. [UY11, prop. 5.1] for example).

Example 4.5. (Semi-abelian varieties)

Any semi-abelian variety admits a canonical semi-algebraic structure generalizing Example 4.3 and Example 4.4 (we leave the details to the reader).

Example 4.6. (Shimura varieties)

Let $S = \Gamma \backslash X^+$ be a connected pure Shimura variety associated to a Shimura datum (\mathbf{G}, X) (with the notations of Section 2). For simplicity we assume that Γ is torsion-free, equivalently that S is smooth (the meticulous reader will easily extend Definition 4.1 and Definition 4.2 to the orbifold case). Hence $\pi: X^+ \longrightarrow S$ is the universal cover of S. Fix a faithful algebraic representation $\rho: \mathbf{G} \longrightarrow \mathbf{GL}(V)$. As X is a $\mathbf{G}(\mathbb{R})$ -conjugacy class of morphisms from \mathbf{S} to $\mathbf{G}_{\mathbb{R}}$, any point $x \in X^+$ defines a morphism $\rho \circ x: \mathbf{S} \longrightarrow \mathbf{GL}(V)$, i.e. a Hodge structure V_x on V. Let F_x^{\bullet} be the corresponding Hodge filtration on $V_{\mathbb{C}}$. The Borel embedding $D: X^+ \longrightarrow \hat{X}$ associates to a point $x \in X^+$ the filtration F_x in the complex algebraic flag variety \hat{X} parametrizing filtrations of $V_{\mathbb{C}}$ of a given type. This is an open holomorphic embedding of X^+ in its dual compact space of X^+ . The flag variety \hat{X} is homogeneous under the algebraic action of $\mathbf{G}^{\mathrm{ad}}(\mathbb{C})$ and the open embedding D is equivariant under the natural inclusion $h: \Gamma \hookrightarrow \mathbf{G}^{\mathrm{ad}}(\mathbb{R})^+ \hookrightarrow \mathbf{G}^{\mathrm{ad}}(\mathbb{C})$, hence (D,h) defines a bi-algebraic structure on S.

The identification of the bi-algebraic varieties for this bi-algebraic structure is due to Ullmo and Yafaev [UY11]:

Theorem 4.7. Let S be a pure connected Shimura variety endowed with its canonical bi-algebraic structure. The bi-algebraic subvarieties of S are the weakly special ones.

Let us indicate the proof of Theorem 4.7, which illustrates typical reduction steps and monodromy arguments.

Let (\mathbf{G}, X) be the Shimura datum defining S (hence S is a connected component of the Shimura variety $\mathrm{Sh}_K(\mathbf{G}, X)$, for some compact open subgroup $K \subset \mathbf{G}(\mathbb{A}_f)$).

Any weakly special subvariety W of S is an algebraic subvariety of S image under $\pi: X^+ \longrightarrow S = \Gamma \backslash X^+$ of a totally geodesic Hermitian subdomain $X^+_{\mathbf{H}} \subset X^+$. As $X^+_{\mathbf{H}}$ is the intersection of the algebraic subvariety $\hat{X}_{\mathbf{H}} \subset \hat{X}$ with X^+ , the weakly special W is bi-algebraic.

Conversely we want to show that any bi-algebraic subvariety of S is weakly special. Let $W \subset S$ be an algebraic subvariety.

- Replacing if necessary S by its smallest special subvariety containing W, we can assume without loss of generality that W is Hodge generic in S.
- The morphism $\psi: \mathbf{G} \longrightarrow \mathbf{G}^{\mathrm{ad}}$ from \mathbf{G} to its adjoint group extends to a morphism of Shimura data $\psi: (\mathbf{G}, X) \longrightarrow (\mathbf{G}^{\mathrm{ad}}, X^{\mathrm{ad}})$. Let $K^{\mathrm{ad}} \subset \mathbf{G}^{\mathrm{ad}}(\mathbb{A}_{\mathrm{f}})$ be a compact open

subgroup containing the image of K. We thus have a morphism of Shimura varieties $\psi: \operatorname{Sh}_K(\mathbf{G}, X) \longrightarrow \operatorname{Sh}_{K^{\operatorname{ad}}}(\mathbf{G}^{\operatorname{ad}}, K^{\operatorname{ad}})$. In this situation one immediately checks that W is weakly special if and only if $\psi(W)$ is weakly special. Moreover as the connected components of X and X^{ad} coincide, W is bi-algebraic if and only if $\psi(W)$ is bi-algebraic. Hence we can assume that \mathbf{G} is adjoint.

- Changing the level if necessary we can also assume without loss of generality that K is sufficiently small so that S is smooth.

Fix a faithful rational representation $\rho: \mathbf{G} \hookrightarrow \mathbf{GL}(V)$ and an integral structure $V_{\mathbb{Z}} \subset V$ such that $\Gamma \subset \mathbf{GL}(V_{\mathbb{Z}})$. This defines a polarized \mathbb{Z} -variation of Hodge structures \mathbb{V} on S. Let $\rho: \pi_1(W^{\mathrm{sm}}) \longrightarrow \Gamma \subset \mathbf{GL}(V_{\mathbb{Z}})$ be the monodromy representation of the induced variation on the smooth locus W^{sm} of W and $\Gamma_W := \rho(\pi_1(W^{\mathrm{sm}}))$. Let $\widetilde{W} \subset X^+$ be an analytic irreducible component of $\pi^{-1}(W)$. Hence the group Γ_W is exactly the stabilizer of \widetilde{W} in Γ .

Suppose from now on that W is bi-algebraic. Hence $\widetilde{W} \subset X^+$ is algebraic of the form $\widehat{W} \cap X^+$, where $\widehat{W} \subset \widehat{X}$ is the Zariski-closure of W in \widehat{X} . In particular \widehat{W} is stabilized by the algebraic monodromy \mathbf{G}_1 , connected component of the Zariski-closure of Γ_W in \mathbf{G} . Recall the following result of Deligne (generalized by André [An92] in the mixed case):

Theorem 4.8. Let \mathbb{V} be an admissible variation of mixed Hodge structures on a smooth quasi-projective variety S with generic Mumford-Tate group G.

- (i) The algebraic monodromy group $G_1 \subset G$ is a normal subgroup of the derived group G^{der} .
- (ii) If moreover S contains a CM-point then $\mathbf{G}_1 = \mathbf{G}^{\mathrm{der}}$.

Applying (i) and as **G** is adjoint, we obtain a decomposition of Shimura data

$$(\mathbf{G}, X) = (\mathbf{G}_1, X_1) \times (G_2, X_2)$$

and one checks that W is the π - image of $X_1^+ \times x_2$ for a Hodge generic point $x_2 \in X_2^+$. If follows from Moonen's Theorem 3.5 that W is weakly special.

The construction of a natural bi-algebraic structure on a pure Shimura variety extends to mixed Shimura varieties, as well as the identification of bi-algebraic subvarieties with weakly special ones (see [Gao16]).

4.2. $\overline{\mathbb{Q}}$ -bi-algebraic geometry. Let S be a complex algebraic variety with a bialgebraic structure as in Section 4.1. While positive dimensional bi-algebraic subvarieties are usually rare and of geometric significance, any point of S is bi-algebraic in the sense of Definition 4.2. To obtain a more meaningful definition of bi-algebraic points we refine Definition 4.1 as follows:

Definition 4.9. A $\overline{\mathbb{Q}}$ -bi-algebraic structure on a complex algebraic variety S is a complex bi-algebraic structure $(D: \tilde{S} \longrightarrow \hat{X}, h: \pi_1(S) \longrightarrow \operatorname{Aut}(\hat{X}))$ such that:

- (1) S is defined over $\overline{\mathbb{Q}}$.
- (2) $\hat{X} = \hat{X}_{\overline{\mathbb{Q}}} \otimes_{\overline{\mathbb{Q}}} \mathbb{C}$ is defined over $\overline{\mathbb{Q}}$ and the homomorphism h takes value in $\operatorname{Aut}_{\overline{\mathbb{Q}}} X_{\overline{\mathbb{Q}}}$.

Definition 4.10. Let (D,h) be a $\overline{\mathbb{Q}}$ -bi-algebraic structure on S. A point $s \in S(\mathbb{C})$ is said to be an arithmetic bi-algebraic point if $s \in S(\overline{\mathbb{Q}})$ and any (equivalently one) π -pre-image $\tilde{s} \in \tilde{S}$ satisfies $D(\tilde{s}) \in \hat{X}_{\overline{\mathbb{Q}}}(\overline{\mathbb{Q}})$.

Let us emphasize that the choice of the $\overline{\mathbb{Q}}$ -structure on \hat{X} and the normalization of the developing map D crucially determines the existence of a large supply of arithmetic bi-algebraic points.

Example 4.11. (Tori)

If we endow \mathbb{C}^n and $(\mathbb{C}^*)^n$ with their standard rational structure \mathbb{Q}^n and $(\mathbb{Q}^*)^n$, the arithmetic bi-algebraic points of $(\mathbb{C}^*)^n$ for the $\overline{\mathbb{Q}}$ -bi-algebraic structure defined in Example 4.3 are exactly the torsion points. Indeed, without loss of generality we can assume n=1. The Gelfond-Schneider theorem [Ge60] states that if α and β are complex numbers such that $\alpha \neq 0$ and e^{α} , β and $e^{\alpha\beta}$ are all in $\overline{\mathbb{Q}}$ then $\beta \in \mathbb{Q}$. Applying to $\alpha = 2\pi i$, we see that $\beta \in \mathbb{C}^*$ is bi-algebraic if and only if it is of the form $\exp(2\pi i\beta)$ with $\beta \in \mathbb{Q}$, i.e. α is a torsion point.

Notice that if we had chosen for the uniformization map the usual exponential exp: $\mathbb{C} \longrightarrow \mathbb{C}^*$ rather than $\exp(2\pi i \cdot) : \mathbb{C} \longrightarrow \mathbb{C}^*$ (keeping the same rational structures $\mathbb{Q} \subset \mathbb{C}$ and $\mathbb{Q}^* \subset \mathbb{C}^*$), or if we had kept the same uniformization map but chosen the rational structure $\mathbb{Q}(1)$ of \mathbb{C} , the only arithmetic bi-algebraic point for \mathbb{C}^* would have been 1 by the Hermite-Lindemann theorem [Ge60].

Example 4.12. (Abelian varieties with CM)

In the setting of Example 4.4, suppose from now on that A is an Abelian variety over $\overline{\mathbb{Q}}$. If we define a $\overline{\mathbb{Q}}$ -bi-algebraic structure on $A_{\mathbb{C}}$ by choosing the standard $\overline{\mathbb{Q}}$ -model Lie $(A_{\overline{\mathbb{Q}}})$ of Lie $(A_{\mathbb{C}})$, the unique bi-algebraic point of $A_{\mathbb{C}}$ is the identity (see [Lang66, thm.3 p.28]).

When A is a complex Abelian variety of dimension g with CM (hence A is in particular defined over $\overline{\mathbb{Q}}$) one can consider a better $\overline{\mathbb{Q}}$ -structure on $\mathrm{Lie}(A_{\mathbb{C}})$: in this case the lattice of periods $\Gamma := \mathrm{Ker}\,\pi \subset \mathrm{Lie}(A)$ generates a $\overline{\mathbb{Q}}$ -vector space $V_{\overline{\mathbb{Q}}} \subset \mathrm{Lie}(A)$ of dimension g, hence defines a $\overline{\mathbb{Q}}$ -structure on $\mathrm{Lie}(A)$. In [Ma76] Masser proved:

Theorem 4.13. (Masser) Let A be a complex Abelian variety of dimension g with CM. Let $V_{\overline{\mathbb{Q}}} \subset \text{Lie}(A)$ be the $\overline{\mathbb{Q}}$ -vector space generated by the lattice of periods Γ . Arithmetic bi-algebraic points for this $\overline{\mathbb{Q}}$ -bi-algebraic structure on A are exactly the torsion points of A.

Example 4.14. (Semi-abelian varieties whose Abelian part has CM) Example 4.11 and Example 4.12 can be combined to define a $\overline{\mathbb{Q}}$ -bi-algebraic structure on any semi-abelian variety whose Abelian part has CM. Once more the arithmetic bi-algebraic points are the torsion points. We leave the details to the reader.

Example 4.15. (Shimura varieties)

Let (\mathbf{G}, X) be a pure Shimura datum and $K \subset \mathbf{G}(\mathbb{A}_f)$ a compact open subgroup. A fundamental result of the theory of Shimura varieties is that the complex quasi-projective variety $\mathrm{Sh}_K(\mathbf{G}, X)$ is defined over a number field $E(\mathbf{G}, X)$ (called the reflex field) depending only on the Shimura datum (\mathbf{G}, X) . It follows that any pure connected Shimura variety $S = \Gamma \backslash X^+$, connected component of $\mathrm{Sh}_K(\mathbf{G}, X)$, is defined over an Abelian extension of $E(\mathbf{G}, X)$.

With the notations of Section 2, the flag variety \hat{X} is naturally defined over \mathbb{Q} as V is. This defines a $\overline{\mathbb{Q}}$ -bi-algebraic structure on S. The arithmetic bi-algebraic points of S for this $\overline{\mathbb{Q}}$ -bi-algebraic structure on S are the points of $S(\overline{\mathbb{Q}})$ whose pre-images lie in $X^+ \cap \hat{X}(\overline{\mathbb{Q}})$. An easy argument given in [UY11, section 3.4] shows that special points are always arithmetic bi-algebraic points.

What about the converse? When $(\mathbf{G}, X) = (\mathbf{GL}_2, \mathcal{H}^{\pm})$ and S is the modular curve $Y(1) \simeq \mathbb{C}$, Schneider's theorem [Schn37] states that if $\tau \in \mathcal{H} \cap \overline{\mathbb{Q}}$ and $x = j(\tau) \in \overline{\mathbb{Q}}$ then τ is imaginary quadratic i.e. x is a CM-point. Hence the bi-algebraic points are exactly the special points.

Cohen [Co96] and Shiga-Wolfart [ShWo95] generalize this result to A_g . A formal argument generalize their result to Shimura varieties of Abelian type:

Theorem 4.16. (Cohen, Shiga, Wolfart) A point $x \in \mathcal{A}_g(\overline{\mathbb{Q}})$ is an arithmetic bi-algebraic point if and only if it is special.

More generally let (\mathbf{G}, X) be a Shimura datum of Abelian type, $K \subset \mathbf{G}(\mathbb{A}_f)$ a compact open subgroup and S a connected component of $\mathrm{Sh}_K(\mathbf{G}, X)$ endowed with the $\overline{\mathbb{Q}}$ -bialgebraic structure defined above. A point of S is bi-algebraic if and only if it is special.

Using Example 4.14, both the definition of a natural $\overline{\mathbb{Q}}$ -bi-algebraic structure and Theorem 4.16 extend to mixed Shimura varieties whose pure part is of Abelian type.

Remark 4.17. It is worth underlining that all numerical transcendence results used to define interesting $\overline{\mathbb{Q}}$ -bi-algebraic structures are subsumed in the fundamental analytic subgroup theorem of Wüstholz [Wus89]:

Theorem 4.18. Let \mathbf{G} be a commutative algebraic group over $\overline{\mathbb{Q}}$ with Lie algebra \mathfrak{g} and $\exp: \mathfrak{g}_{\mathbb{C}} \longrightarrow \mathbf{G}(\mathbb{C})$ its complex exponential map. Let $\mathfrak{b} \subset \mathfrak{g}$ be a $\overline{\mathbb{Q}}$ -vector subspace of positive dimension and $B := \exp(\mathfrak{b} \otimes_{\overline{\mathbb{Q}}} \mathbb{C})$.

Then $B \cap \mathbf{G}(\overline{\mathbb{Q}}) \neq 0$ if and only if there exists a positive dimensional $\overline{\mathbb{Q}}$ -algebraic subgroup $\mathbf{H} \subset \mathbf{G}$ such that $\mathbf{H}(\mathbb{C}) \subset B$.

4.3. Special structures and bi-algebraic structures.

Definition 4.19. A special structure on a complex algebraic variety S is of bi-algebraic origin if S admits a bi-algebraic structure such that the special subvarieties of S are its bi-algebraic subvarieties containing a special point. Such a bi-algebraic structure is said to underly the special structure.

A special structure on a complex algebraic variety S is said to be of \mathbb{Q} -bi-algebraic origin if it admits an underlying $\overline{\mathbb{Q}}$ -bi-algebraic structure whose arithmetic bi-algebraic points are the special points.

Thus the special structures we defined on semi-abelian varieties and mixed Shimura varieties are of bi-algebraic origin. If moreover the Abelian part of the semi-Abelian variety has CM or the pure part of the mixed Shimura variety is of Abelian type, it follows from Example 4.14 and Example 4.15 that the special structure is of $\overline{\mathbb{Q}}$ -bi-algebraic origin.

4.4. The Ax-Lindemann conjecture. In the abstract context of bi-algebraic geometry, the Ax-Lindemann conjecture is the following functional transcendence statement:

Conjecture 4.20. (abstract Ax-Lindemann): Let S be an irreducible algebraic variety endowed with a bi-algebraic structure. For any irreducible algebraic subvariety $Y \subset \tilde{S}$, the Zariski-closure $\overline{\pi(Y)}^{\text{Zar}}$ is a bi-algebraic subvariety of S.

Equivalently: for any algebraic subvariety $V \subset S$, any irreducible algebraic subvariety Y of X contained in $\pi^{-1}(V)$ and maximal for this property is bi-algebraic.

Example 4.21. (semi-Abelian varieties) Ax [Ax71] showed that the abstract Ax-Lindemann conjecture is true for any semi-Abelian variety endowed with the bi-algebraic structure of Example 4.5:

Theorem 4.22. (Ax) Let A be a semi-Abelian variety and $\pi: \mathbb{C}^n \longrightarrow A$ its uniformization. For any irreducible algebraic subvariety $Y \subset \mathbb{C}^n$, the Zariski-closure $\overline{\pi(Y)}^{Zar}$ of $\pi(Y)$ is the translate of an algebraic subgroup of A.

Remark 4.23. Notice that Ax's theorem for $\pi := (\exp(2\pi i \cdot), \dots, \exp(2\pi i \cdot)) : \mathbb{C}^n \longrightarrow (\mathbb{C}^*)^n$ is the functional analog of the classical Lindemann transcendence theorem stating that if $\alpha_1, \dots, \alpha_n$ are \mathbb{Q} -linearly independent algebraic numbers then $e^{\alpha_1}, \dots, e^{\alpha_n}$ are algebraically independent over \mathbb{Q} . This explain our terminology.

Example 4.24. (Shimura variety)

Theorem 4.25. (Hyperbolic Ax-Lindemann) Let $\pi: X \longrightarrow S$ be the uniformization map of a connected mixed Shimura variety. We endow S with the bi-algebraic structure of Example 4.6.

- (i) For any irreducible algebraic subvariety $Y \subset X$, the Zariski-closure $\overline{\pi(Y)}^{\operatorname{Zar}}$ of $\pi(Y)$ is weakly special.
- (ii) Equivalently, let W be an algebraic subvariety of S. Irreducible algebraic subvarieties of X contained in $\pi^{-1}W$ and maximal for this property are precisely the irreducible components of the pre-images of maximal weakly special subvarieties of S contained in W.

5. O-MINIMAL GEOMETRY AND THE PILA-WILKIE'S THEOREM

5.1. **O-minimal structures.** For a more detailed treatment of o-minimality we refer to [vdD98], [PW06], [PetStar10], [Pil] and [Sca16].

Definition 5.1. A structure **S** is a collection $\mathbf{S} = (S_n)_{n \in \mathbb{N}}$, where S_n is a set of subsets of \mathbb{R}^n , called the definable sets of the structure, such that for every $n \in \mathbb{N}$:

- (1) all algebraic subsets of \mathbb{R}^n are in S_n .
- (2) S_n is a boolean subalgebra of the power set of \mathbb{R}^n .
- (3) If $A \in S_n$ and $B \in S_m$ then $A \times B \in S_{n+m}$.
- (4) Let $p: \mathbb{R}^{n+1} \longrightarrow \mathbb{R}^n$ be a linear projection. If $A \in S_{n+1}$ the $p(A) \in S_n$.

A function $f: \mathbb{R}^n \longrightarrow \mathbb{R}^n$ is said to be definable if its graph is.

A dual point of view starts from the functions, namely considers sets definable in a first-order structure

$$\langle \mathbb{R}, +, \times, <, (f_i)_{i \in I} \rangle$$

where I is a set and the $f_i : \mathbb{R}^{n_i} \longrightarrow \mathbb{R}$, $i \in I$, are functions. A subset $Z \subset \mathbb{R}^n$ is definable if it can be defined by a formula

$$Z := \{(x_1, \dots, x_n) \in \mathbb{R}^n / \phi(x_1, \dots, x_n) \text{ is true} \}$$

where ϕ is a first-order formula that can be written using only the quantifiers \forall and \exists applied to real variables, logical connectors, algebraic expressions written with the f_i 's, < and fixed parameters $\lambda_i \in \mathbb{R}$. When the set I is empty the definable subsets are the semi-algebraic sets. Semi-algebraic subsets are thus always definable.

The o-minimal axiom for a structure S guarantees the possibility of doing geometry using definable sets as basic blocks. In particular it excludes Cantor sets from S:

Definition 5.2. A structure **S** is said to be o-minimal if the definable subsets of \mathbb{R} are precisely the finite unions of points and intervals (i.e. the semi-algebraic subsets of \mathbb{R}).

Example 5.3. The structure $\mathbb{R}_{\sin} := \langle \mathbb{R}, +, \times, <, \sin \rangle$ is not o-minimal. Indeed the infinite union of points $\pi \mathbb{Z} = \{x \in \mathbb{R}, \sin x = 0\}$ is a definable subset of \mathbb{R} in this structure.

A deep theorem of Wilkie [Wil96] states:

Theorem 5.4. The structure $\mathbb{R}_{exp} := \langle \mathbb{R}, +, \times, <, \exp \rangle$ is o-minimal.

Definition 5.5. A function $f: \mathbb{R}^n \longrightarrow \mathbb{R}$ is a restricted analytic function if it is zero outside $[0,1]^n$ and if there exists a real analytic function g on a neighbourhood of $[0,1]^n$ such that f and g are equal on $[0,1]^n$.

One defines $\mathbb{R}_{an} := \langle \mathbb{R}, +, \times, <, \{f\} \text{ for } f \text{ restricted analytic function} \rangle$.

A theorem of Van den Dries based on Gabrielov's results [Ga68] shows:

Theorem 5.6. The structure \mathbb{R}_{an} is o-minimal.

In diophantine geometry we will use the structure

$$\mathbb{R}_{\text{an,exp}} := \langle \mathbb{R}, +, \times, <, \exp, \{f\} \text{ for } f \text{ restricted analytic function} \rangle$$

generated by \mathbb{R}_{an} and \mathbb{R}_{exp} . The structure generated by two o-minimal structures is not o-minimal in general, but Van den Dries and Miller [vdDM85] prove in this case:

Theorem 5.7. The structure $\mathbb{R}_{an,exp}$ is o-minimal.

5.2. Pila-Wilkie's counting theorem. Let H denote the standard multiplicative height function on $\overline{\mathbb{Q}}$. Thus if L is a number field, M_L its set of places, and $x \in L$ then

$$H(x) := \prod_{v \in M_L} \max(1, |x|_v) .$$

We also denote by H its extension to $\overline{\mathbb{Q}}^n$ defined by $H(x_1, \dots, x_n) := \max_i H(x_i)$. Given a subset $Z \subset \mathbb{R}^n$, a positive integer d and a real number T we define

(5.1)
$$\Theta_d(Z,T) := \{(x_1, \dots, x_n) \in Z \mid \max_i [\mathbb{Q}(x_i) : \mathbb{Q}] \leq d \text{ and } H(x_1, \dots, x_n) \leq T \}$$
, and

$$(5.2) N_d(Z,T) := |\Theta_d(Z,T)| .$$

Definition 5.8. Let $Z \subset \mathbb{R}^n$. We denote by Z^{alg} the union of all positive dimensional semi-algebraic subsets of \mathbb{R}^n contained in Z.

Theorem 5.9. (Pila-Wilkie [PW06]) Let $Z \subset \mathbb{R}^n$ be a subset definable in an o-minimal expansion of \mathbb{R} (typically: $\mathbb{R}_{an,exp}$). Let d be a positive integer and ε a positive real number. There exists a constant $c = c(Z, d, \varepsilon)$ such that

(5.3)
$$\forall T > 0, \quad N_d(Z - Z^{\text{alg}}, T) \le c \cdot T^{\varepsilon} .$$

In particular if there exists $\alpha > 0$ and c' = c'(d, Z) > 0 such that for any T sufficiently large we have $N_d(Z, T) \geq c' \cdot T^{\alpha}$ then Z^{alg} is non-empty.

Example 5.10. Let $Z \subset I^2$ be the intersection of a real analytic curve C defined in a neighborhood of I^2 with I^2 (where I = [0,1]). Hence Z definable in \mathbb{R}_{an} . Suppose that there exist a positive integer d, and real numbers $\alpha > 0$ and c' = c'(d, Z) > 0 such that for any T sufficiently large we have $N_d(Z,T) \geq c' \cdot T^{\alpha}$. Then the real analytic curve C is real algebraic.

Theorem 5.9 can be refined in two directions, which are used in the Pila-Zannier strategy. The first refinement uses the notion of semi-algebraic bloc.

Definition 5.11. A semi-algebraic bloc W in \mathbb{R}^n for an o-minimal expansion of \mathbb{R} is a connected definable subset of \mathbb{R}^n , regular at every point, such that there exists a connected positive dimensional semi-algebraic set $B \subset \mathbb{R}^n$ containing W and which coincide with W in the neighbourhood of every point of W. In particular a semi-algebraic bloc is covered by open semi-algebraic sets.

Example 5.12. Let $W := \{(x, y) \in \mathbb{R}^2, y < \exp(x)\}$. This is a semi-algebraic bloc of \mathbb{R}_{\exp} with $B = \mathbb{R}^2$.

Theorem 5.13. Let $Z \subset \mathbb{R}^n$ be a subset definable in an o-minimal expansion of \mathbb{R} . Let d be a positive integer and ε a positive real number. There exists a constant $c = c(Z, d, \varepsilon)$ such that $\Theta_d(Z,T)$ is contained in at most $c \cdot T^{\varepsilon}$ semi-algebraic blocs contained in Z.

The second refinement deals with families.

Definition 5.14. A definable family $Z := \{Z_b\}_{b \in B}$ of subsets of \mathbb{R}_n is a definable subset of $\mathbb{R}^n \times \mathbb{R}^m$ whose projection on the second factor is $B \subset \mathbb{R}^m$.

In this case every fiber $Z_b \subset \mathbb{R}^n$ for $b \in B$ is definable.

Theorem 5.15. Let $Z := \{Z_b\}_{b \in B}$ be a definable family of subsets of \mathbb{R}^n in an o-minimal expansion of \mathbb{R} . Let ε be a positive real number. There exists a constant $c := c(\varepsilon, Z)$ and a definable family $Y := \{Y_b\}_{b \in B}$ of subsets of \mathbb{R}^n such that, for every $b \in B$, one has the inclusion $Y_b \subset Z_b^{\text{alg}}$ and

$$(5.4) N_d(Z_b - Y_b, T) \le c \cdot T^{\varepsilon} .$$

Remarks 5.16. (a) The crux of this refinement is the uniformity (the constant c does not depend of $b \in B$).

(b) The use of the definable family $\{Y_b\}_{b\in B}$ is needed as the set Z^{alg} associated to a definable set Z is usually not definable. Consider for example [Sca16, Rem. 4.5] the \mathbb{R}_{exp} -definable subset of \mathbb{R}^3 defined as

$$Z := \{(x, y, z) \in \mathbb{R}^3_+ \; ; \; z = x^y \}$$

whose algebraic part Z^{alg} is the union of triples $(x, y, z) \in Z$ such that $y \in \mathbb{Q}$.

The proof of Theorem 5.9 and its refinements relies on a reparametrization theorem generalizing a result of Gromov and Yomdin for semi-algebraic sets:

Theorem 5.17. Let r be an integer. Let $Z \subset (0,1)^n$ be a definable set in an o-minimal expansion of \mathbb{R} , of dimension m. There exists a finite set I := I(Z,r), uniformly bounded when Z varies in a definable family, such that

$$Z = \bigcap_{i \in I} \phi_i((0,1)^m)$$

where $\phi_i: (0,1)^m \longrightarrow (0,1)^n$ is of class C^r and $|\partial_{\alpha}\phi_i| \leq 1$ for any multi-index α of length $|\alpha| \leq r$.

6. O-MINIMALITY AND SHIMURA VARIETIES

We will not pursue here how to use o-minimality in the general context of special structures of bi-algebraic origin. From now on we restrict ourselves to the context of (mixed) Shimura varieties.

6.1. Definability of π restricted to a fundamental set. Let $\pi: X^+ \longrightarrow S := \Gamma \backslash X^+$ be the uniformization of a connected mixed Shimura variety S. The realization $X^+ \subset \hat{X}$ defines X^+ as a real semi-algebraic subset of \hat{X} . Of course the map π cannot be definable in any o-minimal structure as it is periodic under the infinite group Γ . We remove this difficulty by restricting π to a fundamental set of X^+ for the action of Γ :

Definition 6.1. A fundamental set for the action of Γ on X^+ is a connected open subset \mathcal{F} of X^+ such that $\Gamma \overline{\mathcal{F}} = X^+$ and such that the set $\{\gamma \in \Gamma \mid \gamma \mathcal{F} \cap \mathcal{F} \neq \emptyset\}$ is finite.

An essential step for using o-minimal geometry in the context of Shimura varieties is the following result:

Theorem 6.2. There exists a semi-algebraic fundamental set \mathcal{F} for the action of Γ on X^+ such that the restriction $\pi_{|\mathcal{F}} \colon \mathcal{F} \longrightarrow S$ is definable in the o-minimal structure $\mathbb{R}_{\mathrm{an,exp}}$.

The special case of Theorem 6.2 when S is pure and compact is easy, see [UY14b, Prop.4.2]. In this case, the map $\pi_{|\mathcal{F}}$ is even definable in \mathbb{R}_{an} . Theorem 6.2 in the case where $X = \mathbf{H}_g$ is the Siegel upper half plane of genus g was proven by Peterzil and Starchenko (see [PetStar13] and [PetStar10]): in this case they use an explicit description for π in terms of θ -functions and delicate computations with these. Notice moreover that this particular case implies Theorem 6.2 for any special subvariety S of \mathcal{A}_g (see Proposition 2.5 of [U14]). On the other hand Peterzil and Starchenko's method does not generalize to general arithmetic varieties, where an explicit description of π is not available. The paper [KUY16] provides a completely geometric proof of Theorem 6.2 for any pure Shimura variety using the general theory of toroidal compactifications of arithmetic varieties (cf. [AMRT75]). Gao generalizes this result to mixed Shimura varieties in [Gao16].

Let us give the proof of Theorem 6.2 in the baby-case of S = Y(1) and $\pi = j : \mathbf{H} \longrightarrow Y(1) = \mathbf{SL}(2,\mathbb{Z}) \backslash \mathbf{H} \simeq \mathbb{C}$. In this case we consider for \mathcal{F} the usual semi-algebraic

fundamental set:

(6.1)
$$\mathcal{F} := \{ z = x + iy \in \mathbf{H}, -\frac{1}{2} < x < \frac{1}{2} \text{ and } y > \frac{\sqrt{3}}{2} \}.$$

Let us consider the diagram of holomorphic maps:

$$\mathcal{F} \subset \mathbf{H} \xrightarrow{z \mapsto e^{2\pi i z}} \Delta^* \xrightarrow{q} S = \mathbb{C}$$
,

where $\Delta^* := \{z \in \mathbb{C}^*, |z| < \exp(-\pi\sqrt{3})\}$. We claim that this composite is definable in $\mathbb{R}_{an,exp}$. It follows from the following observations:

- $\exp(2\pi i z) = \exp(-2\pi \operatorname{Im}(z)) \cdot \exp(2\pi i \operatorname{Re}(z))$. The first factor is definable in \mathbb{R}_{\exp} . On the other hand $\operatorname{Re}(x)$ is bounded on \mathcal{F} , hence the second factor restricted to \mathcal{F} is definable in $\mathbb{R}_{\operatorname{an}}$.

- The function $q: \Delta^* \longrightarrow \mathbb{C}$ extends to $\Delta \longrightarrow \mathbb{P}^1\mathbb{C}$ hence is definable in \mathbb{R}_{an} .

For a general pure connected Shimura variety S associated with a Shimura datum (\mathbf{G}, X) , the fundamental set \mathcal{F} is a semi-algebraic Siegel set, whose construction we recall now (see [Bor69] for a general reference). Without loss of generality we can assume that \mathbf{G} is semi-simple of adjoint type. Let \mathbf{P} be a minimal \mathbb{Q} -parabolic subgroup of \mathbf{G} and $K_{\infty} \subset \mathbf{G}(\mathbb{R})$ a maximal compact subgroup such that $K_{\infty} \cap \mathbf{P}(\mathbb{R})$ is a maximal compact subgroup of $\mathbf{P}(\mathbb{R})$. Let \mathbf{U} be the unipotent radical of \mathbf{P} and let \mathbf{A} be a maximal split torus of \mathbf{P} . We denote by \mathbf{S} a maximal split torus of $\mathbf{GL}(V)$ containing $\rho(\mathbf{A})$, by \mathbf{M} the maximal anisotropic subgroup of the connected centralizer $\mathbf{Z}(\mathbf{A})^0$ of \mathbf{A} in \mathbf{P} and by Δ the set of positive simple roots of \mathbf{G} with respect to \mathbf{A} and \mathbf{P} . We denote by $A \subset \mathbf{S}(\mathbb{R})$ the real torus $\mathbf{A}(\mathbb{R})$. For any real number t > 0 we let

$$A_t := \{ a \in A \mid a^{\alpha} \ge t \text{ for any } \alpha \in \Delta \} .$$

A Siegel set for $\mathbf{G}(\mathbb{R})$ for the data $(K_{\infty}, \mathbf{P}, \mathbf{A})$ is a product:

$$\Sigma'_{t,\Omega} := \Omega \cdot A_t \cdot K_{\infty} \subset \mathbf{G}(\mathbb{R})$$

where Ω is a compact neighborhood of e in $\mathbf{M}^0(\mathbb{R}) \cdot \mathbf{U}(\mathbb{R})$.

The image

$$\Sigma_{t,\Omega} := \Omega \cdot A_t \cdot x_0 \subset X^+$$

of $\Sigma'_{t,\Omega}$ in X^+ (where x_0 is the point of $X^+ = \mathbf{G}(\mathbb{R})/K_{\infty}$ fixed under K_{∞}) is called a Siegel set in X^+ .

Theorem 6.3. There exist a semi-algebraic Ω , a real number t and a finite subset J of $\mathbf{G}(\mathbb{Q})$ such that $\mathcal{F} := J \cdot \Sigma_{t_0,\Omega}$ is a fundamental set for the action of Γ on X^+ satisfying Theorem 6.2.

6.2. Heights of special points. In Example 4.15 we define an $\overline{\mathbb{Q}}$ -bi-algebraic structure on any Shimura variety S whose pure part is of Abelian type: special points are exactly the arithmetic bi-algebraic points. A crucial ingredient for applying the Pila-Wilkie's Theorem 5.9 in this context consists in showing that for any special point $x \in S$, the fiber $\pi^{-1}(x)$ consists of algebraic points of X^+ defined over extensions of uniformly bounded degree over \mathbb{Q} ; moreover one controls the height of points of $\pi^{-1}(x) \cap \mathcal{F}$. For simplicity let us state the result for $S = \mathcal{A}_g$ (the first part is classical, the second is due to Pila and Tsimerman):

- **Theorem 6.4.** (1) The uniformization $\pi : \mathbf{H}_g \longrightarrow \mathcal{A}_g = \mathbf{Sp}(2g, \mathbb{Z}) \backslash \mathbf{H}_g$ can be normalized in such a way that the coordinates of the inverse images by π of CM-points of \mathcal{A}_g lie in algebraic extensions of uniformly bounded degree.
 - (2) One can choose the fundamental set \mathcal{F} in Theorem 6.2 for the action of $\mathbf{Sp}(2g,\mathbb{Z})$ on \mathbf{H}_g , and positive real numbers $\alpha = \alpha(g)$ and c = c(g) such that if $x \in \mathcal{A}_g$ is a CM-point parametrizing the Abelian variety A_x and if $\tilde{x} \in \mathcal{F} \cap \pi^{-1}(x)$ then

$$H(\tilde{x}) \le c \cdot d_x^{\alpha}$$
,

where H denotes the canonical height on $M_g(\overline{\mathbb{Q}}) \cap \mathbf{H}_g \subset \overline{\mathbb{Q}}^{n^2}$ and d_x is the absolute value of the discriminant of the center of the ring of endomorphisms of A_x .

Let us write explicitly the case of Y(1). Let $\tau \in \mathcal{F}$ where \mathcal{F} denotes the fundamental set defined in Equation (6.1). If the elliptic curve $E_{\tau} = \mathbb{C}/(\mathbb{Z} \oplus \mathbb{Z}\tau)$ has complex multiplication then τ satisfies a reduced equation $aX^2 + bX + c = 0$ for integers a, b and c such that $|b| \leq a \leq c$. In particular the coordinates of

$$\tau = -\frac{b}{2a} + i\frac{\sqrt{4ac - b^2}}{2a}$$

lie in extensions of degree at most 2 of \mathbb{Q} . Moreover End $(E_{\tau}) = \mathbb{Z}[\tau]$ and the absolute value d_{τ} of the discriminant of End (E_{τ}) is $4ac - b^2$. With our conventions on the height:

$$H(\tau) = \max(H(\frac{b}{2a}), H(\frac{\sqrt{4ac - b^2}}{2a})).$$

On the one hand $H(\frac{b}{2a}) = \max(|b|, 2|a|) = 2|a| \le d_{\tau}$. On the other hand $\frac{\sqrt{4ac-b^2}}{2a}$ is a root of the integral polynomial $4a^2X^2 - d_{\tau}$ hence:

$$H(\frac{\sqrt{4ac-b^2}}{2a}) \le \max(4a^2, d_\tau) \le \frac{4}{3}d_\tau$$
,

where the last inequality follows by noticing that

$$3a^2 \le 4ac - b^2 = d_\tau$$

in view of the inequalities satisfied by (a, b, c).

Finally we obtain $H(\tau) \leq \frac{4}{3}d_{\tau}$.

7. Strategy of the proof of the André-Oort conjecture for \mathcal{A}_q

For $x \in \mathcal{A}_g$ we denote by A_x the principally polarized Abelian variety parametrized by x and d_x the absolute value of the discriminant of the center of the ring of endomorphisms of A_x . We denote by $\pi: \mathbf{H}_g \longrightarrow \mathbf{Sp}(2g, \mathbb{Z}) \backslash \mathbf{H}_g = \mathcal{A}_g$ the uniformization map.

There are two main steps in the proof of Theorem 1.4:

Theorem 7.1. Let $W \subset A_g$ be an algebraic subvariety. There exists a constant C = C(g, W) with the following property. Let x be a special point of A_g contained in W. If $d_x \geq C$ then there exists a positive dimensional special subvariety Z_x of A_g contained in W and containing x.

As there exist only finitely many special points x with d_x smaller than a given constant, Theorem 7.1 implies that there are only finitely many special subvarieties of \mathcal{A}_g contained in W and maximal for these properties which are points. Notice that this proves Theorem 1.4 if W is a curve.

The second step is proven in [U14]:

Theorem 7.2. Let W be a Hodge generic subvariety of a pure connected Shimura's variety S. We assume that if $S = S_1 \times S_2$ is a product of connected Shimura varieties then W is not of the form $W = S_1 \times W_2$ for a special subvariety W' of S_2 .

Then the set of weakly special positive dimensional subvarieties contained in W is not Zariski-dense in W. In particular the set of positive dimensional special subvarieties contained in W is not Zariski dense in W.

7.1. **Proof of Theorem 7.1.** The main ingredient in the proof of Theorem 7.1 is the following result, due to Tsimerman [Tsi] and based on the results of Andreatta-Goren-Howard-Madapusi Pera [AGHM] and Yuan-Zhang [YuZh] on the Colmez conjecture:

Theorem 7.3. There exist positive constants $c_1 = c_1(g)$ and β such that for any special point x of A_g one has:

(7.1)
$$|\operatorname{Gal}(\overline{\mathbb{Q}}/\mathbb{Q}) \cdot x| = [\mathbb{Q}(x) : \mathbb{Q}] \ge c_1 d_x^{\beta}.$$

We will detail Theorem 7.3 in Section 9.4. For now let us show how Theorem 7.3 and o-minimal technics imply Theorem 7.1.

Let $W \subset \mathcal{A}_g$ be an algebraic subvariety defined over $\overline{\mathbb{Q}}$. Replacing W by the union of its conjugate under $\operatorname{Gal}(\overline{\mathbb{Q}})/\mathbb{Q}$) we can assume without loss of generality that W is defined over \mathbb{Q} .

Let $\mathcal{F} \subset \mathbf{H}_g$ be a semi-algebraic fundamental set for the action of $\mathbf{Sp}(2g,\mathbb{Z})$ on \mathbf{H}_g such that $\pi_{|\mathcal{F}}: \mathcal{F} \longrightarrow \mathcal{A}_g$ is definable in $\mathbb{R}_{\mathrm{an,exp}}$ (see Theorem 6.2). Hence the set $\widetilde{W_{\mathcal{F}}} := \pi^{-1}(W) \cap \mathcal{F}$ is definable in $\mathbb{R}_{\mathrm{an,exp}}$.

Let $x \in V$ be a special point. Notice that for any $\sigma \in \operatorname{Gal}(\overline{\mathbb{Q}}/\mathbb{Q})$, $d_{\sigma \cdot x} = d_x$. It follows from Theorem 6.4 that any point y in $\pi^{-1}(\operatorname{Gal}(\overline{\mathbb{Q}}/\mathbb{Q}) \cdot x) \cap \widetilde{W}_{\mathcal{F}}$ is defined in an extension of \mathbb{Q} of uniformly bounded degree and satisfies

$$(7.2) H(y) \le c \cdot d_x^{\alpha} .$$

It follows from Pila-Wilkie Theorem 5.9 and the inequalities (7.1) and (7.2) that if d_x is sufficiently large, there exists a semi-algebraic subset $Y \subset W_{\mathcal{F}}$ of positive dimension, containing one point y in $\pi^{-1}(\operatorname{Gal}(\overline{\mathbb{Q}}/\mathbb{Q}) \cdot x)$. Let Z be an irreducible algebraic subvariety of \mathbf{H}_g contained in $\pi^{-1}(W)$, containing y, and maximal for these properties. Hence Z is positive dimensional. Moreover it follows from the Ax-Lindemann Theorem 4.25 that $\pi(Z)$ is a special subvariety of \mathcal{A}_g contained in V and containing a Galois conjugate $\sigma \cdot x$ of x. As V is defined over \mathbb{Q} the positive dimensional special subvariety $\sigma^{-1}(\pi(Z))$ of \mathcal{A}_g is contained in V and contains x.

7.2. **Proof of Theorem 7.2.** We follow [U16]. Let $\mathcal{E}(W)$ be the set of weakly special subvarieties contained in W. For a positive integer r we denote by $\mathcal{E}_r(W)$ the subset

of $\mathcal{E}(W)$ consisting of weakly special subvarieties of real dimension r. Let d be the maximum r such that $\mathcal{E}_r(W)$ is non-empty.

It follows from the description of weakly special subvarieties that there exists a semi-simple group \mathbf{H} of $\mathbf{G}_{\mathbb{R}}$ and $z_0 \in \mathcal{F}$ such that $\pi(\mathbf{H}(\mathbb{R})) \cdot z_0$ is a weakly special subvariety of W of dimension d. Without loss of generality we can assume that \mathbf{H} has no compact simple real factor.

Let us define

$$B_{\mathbf{H}} := \{(t, z) \in \mathbf{G}(\mathbb{R}) \times \mathcal{F}, \quad \pi(t\mathbf{H}(\mathbb{R})^+ t^{-1} \cdot z) \subset V\}.$$

It follows from analytic continuation that $B_{\mathbf{H}}$ can also be described as:

$$B_{\mathbf{H}} := \{ (t, z) \in \mathbf{G}(\mathbb{R}) \times \mathcal{F}, \quad \pi_{|\mathcal{F}}(t\mathbf{H}(\mathbb{R})^+ t^{-1} \cdot z) \subset V \}.$$

As $\pi_{|\mathcal{F}}$ is definable in $\mathbb{R}_{an,exp}$ (see Theorem 6.2) it follows that $B_{\mathbf{H}}$ is a definable subset of $\mathbf{G}(\mathbb{R}) \times \mathcal{F}$.

Lemma 7.4. Let $(t, z) \in B_{\mathbf{H}}$. Then $\pi(t\mathbf{H}(\mathbb{R})^+t^{-1} \cdot z)$ is a weakly special subvariety of W.

Proof. Let $(t, z) \in B_{\mathbf{H}}$. It follows from the definition of $B_{\mathbf{H}}$ that $t\mathbf{H}(\mathbb{R})^+t^{-1} \cdot z$ is a semi-algebraic subset of X^+ whose projection $\pi(t\mathbf{H}(\mathbb{R})^+t^{-1} \cdot z)$ is contained in W. On the other hand the real dimension of $t\mathbf{H}(\mathbb{R})^+t^{-1} \cdot z$ is at least the dimension of $\mathbf{H}(\mathbb{R})^+ \cdot z$, with equality if and only if $\mathrm{Stab}_{\mathbf{G}(\mathbb{R})} \cap t\mathbf{H}(\mathbb{R})^+t^{-1}$ is a maximal compact subgroup of $t\mathbf{H}(\mathbb{R})^+t^{-1}$.

Let Y be an irreducible algebraic subvariety of X^+ , containing $t\mathbf{H}(\mathbb{R})^+t^{-1}\cdot z$, such that $\pi(Y)\subset W$, and maximal for these properties. By the Ax-Lindemann Theorem 4.25, $\pi(Y)$ is weakly special. It follows from the definition of d that:

$$\dim(\pi(Y)) \le d = \dim(\mathbf{H}(\mathbb{R})^+ \cdot z_0) \le \dim(t\mathbf{H}(\mathbb{R})^+ t^{-1} \cdot z) \le \dim(\pi(Y)) .$$

Hence $\pi(Y) = \pi(t\mathbf{H}(\mathbb{R})^+t^{-1}\cdot z)$, hence $\pi(t\mathbf{H}(\mathbb{R})^+t^{-1}\cdot z)$ is weakly special.

Lemma 7.5. The set $C(\mathbf{H}, W)$ of conjugacy classes $t\mathbf{H}(\mathbb{R})^+t^{-1}$, $t \in \mathbf{G}(\mathbb{R})$, for which there exists $z \in \mathcal{F}$ satisfying $\pi(T\mathbf{H}(\mathbb{R})^+t^{-1} \cdot z) \subset W$, is finite.

Proof. Consider the map $\psi: B_{\mathbf{H}} \longrightarrow \mathbf{G}(\mathbb{R})/N_{\mathbf{G}(\mathbb{R})}(\mathbf{H}(\mathbb{R})^+)$ deduced from the projection on the first factor. Hence $C(\mathbf{H}, W)$ is in bijection with $\psi(B_{\mathbf{H}})$. As $B_{\mathbf{H}}$ is definable and ψ is algebraic, the image $\psi(B_{\mathbf{H}})$ is definable. Moreover if $(t, z) \in B_{\mathbf{H}}$ then $\pi(t\mathbf{H}(\mathbb{R})^+t^{-1}\cdot z)$ is weakly special by Lemma 7.4. From the description of weakly special subvarieties there exists a \mathbb{Q} -algebraic subgroup $\mathbf{H}_{t,\mathbb{Q}} \subset \mathbf{G}$ such that $t\mathbf{H}_{\mathbb{R}}t^{-1} = \mathbf{H}_{t,\mathbb{R}}^{\mathrm{nc}}$. As the set of \mathbb{Q} -algebraic subgroup of \mathbf{G} is countable, it follows that $C(\mathbf{H}, B)$ is countable. Any countable set definable in some o-minimal structure is finite hence $C(\mathbf{H}, W)$ is finite. \square

Lemma 7.6. Under the hypotheses of Theorem 7.2 the union $\bigcup_{V \in \mathcal{E}_d(W)} V$ of the weakly special subvarieties contained in W of maximal dimension d is not Zariski-dense in W.

Proof. As $\mathbf{G}_{\mathbb{R}}$ has only finitely many conjugacy classes of semi-simple subgroups, there exists only finitely many (up to $\mathbf{G}(\mathbb{R})$ -conjugacy) subgroups $\mathbf{H}_{\mathbb{R}}$ of $\mathbf{G}_{\mathbb{R}}$ for which there exists $z_0 \in \mathcal{F}$ with $\pi(\mathbf{H}(\mathbb{R})^+ \cdot z_0) \in \mathcal{E}_d(W)$ and such that $\mathbf{H}_{\mathbb{R}} = \mathbf{H}_{\mathbb{R}}^{\mathrm{nc}}$.

For such an $\mathbf{H}_{\mathbb{R}}$, there exists a semi-simple subgroup $\mathbf{H}_{\mathbb{Q}} \subset \mathbf{G}$ satisfying $(\mathbf{H}_{\mathbb{Q}})^{\mathrm{nc}}_{\mathbb{R}} = \mathbf{H}_{\mathbb{R}}$ and the number of such $\mathbf{H}_{\mathbb{Q}}$ is finite by Lemma 7.5.

Let $\mathbf{H}_{\mathbb{Q}}$ be such a subgroup.

If $\mathbf{H}_{\mathbb{Q}}$ is a factor of \mathbf{G} then S decomposes as $S_1 \times S_2$. Any weakly special subvarieties of the form $\pi(\mathbf{H}_{\mathbb{R}})^+ \cdot z$) with $z \in \mathcal{F}$ is of the form $S_1 \times \{x_2\}$ for some $x_2 \in S_2$. The Zariski-closure of the union of weakly special subvarieties V of the form $\pi(\mathbf{H}_{\mathbb{R}})^+ \cdot z$) is $S_1 \times W'$, where W' denotes the Zariski-closure of the set of x_2 for which $S_1 \times \{x_2\} \subset W$. As W is not of the form $S_1 \times W'$, this union is not Zariski-dense in W.

If $\mathbf{H}_{\mathbb{O}}$ is not normal in \mathbf{G} , one shows the following:

Proposition 7.7. Suppose $\mathbf{H}_{\mathbb{Q}}$ is not normal in \mathbf{G} . Then the union of weakly special subvarieties of the form $\pi(\mathbf{H}_{\mathbb{R}})^+ \cdot z$) is contained in a finite union $\bigcup_{1 \leq i \leq r} V_i$ of strict special subvarieties V_i of S.

As W is Hodge generic, the intersection $W \cap \bigcup_{1 \leq i \leq r} V_i$ is not Zariski-dense in W. This finishes the proof of Lemma 7.6.

One concludes the proof of Theorem 7.2 by induction on the dimension of the weakly special subvarieties of S contained in W. Let us indicate the argument.

Let $d_1 < d$ be the maximal dimension of a weakly special subvariety of W not contained in $\mathcal{E}_d(W)$. There exist a semi-simple subgroup $\mathbf{H}_1 = \mathbf{H}_1^{\mathrm{nc}}$ of $\mathbf{G}_{\mathbb{R}}$ and $z_1 \in \mathcal{F}$ such that $\pi(\mathbf{H}_1(\mathbb{R})^+ \cdot z_1) \subset W$ is of dimension d_1 and is not in $\mathcal{E}_d(W)$. Up to $\mathbf{G}(\mathbb{R})$ -conjugacy there are only finitely any possibilities fo \mathbf{H}_1 . The proof of Lemma 7.4 shows that if $(z,t) \in B_{\mathbf{H}_1}$ and if $\pi(t\mathbf{H}_1(\mathbb{R})^+t^{-1}\cdot z)$ is not contained in $\mathcal{E}_d(W)$ then $\pi(t\mathbf{H}_1(\mathbb{R})^+t^{-1}\cdot z)$ is weakly special contained in W. The proof of Lemma 7.5 shows that the set $C(\mathbf{H}_1,W,\mathcal{E}_d(W))$ of conjugacy classes $t\mathbf{H}_1(\mathbb{R})^+t^{-1}$, $t\in \mathbf{G}(\mathbb{R})$, such that there exists $z\in \mathcal{F}$ with $\pi(t\mathbf{H}_1(\mathbb{R})^+t^{-1}\cdot z)\subset W$ and $\pi(t\mathbf{H}_1(\mathbb{R})^+t^{-1}\cdot z)$ does not belong to $\mathcal{E}_d(W)$, is finite. As in the proof of Lemma 7.6 one concludes that the set of weakly special subvarieties of W of dimension at leat d_1 is not Zariski-dense in W.

By decreasing induction on r one concludes $\bigcup_{r\geq 0}\bigcup_{V\in\mathcal{E}_r(W)}V$ is not Zariski-dense in W.

8. The hyperbolic Ax-Lindemann conjecture

In this section we give indications on the proof of the hyperbolic Ax-Lindemann Theorem 4.25.

8.1. Equivalence of (i) and (ii). Let us first show that the two statements (i) and (ii) of Theorem 4.25 are indeed equivalent.

We first assume (ii). Let Y be an irreducible algebraic subvariety of X^+ . Let W be the Zariski-closure of $\pi(Y)$. Let Z be an irreducible algebraic subvariety of $\pi^{-1}(W)$ containing Y, and maximal for these properties. By hypothesis, $\pi(Z)$ is weakly special, in particular $\pi(Z)$ is irreducible algebraic. As $\pi(Y) \subset \pi(Z) \subset W$, it follows that $\pi(Z) = W$, hence W is weakly special.

Conversely let us assume (i). Let W be an algebraic subvariety of S and Y an irreducible algebraic subvariety of $\pi^{-1}(W)$, maximal for these properties. By hypothesis the Zariski-closure W' of $\pi(Y)$ is weakly special. As $W' \subset W$, there exists an analytic irreducible component Y' of $\pi^{-1}(W')$ containing Y. As W' is weakly special, Y' is

irreducible algebraic. By maximality of Y, one obtains Y = Y' and $\pi(Y) = W'$ is weakly special.

8.2. Stabilizers of maximal algebraic subvarieties of $\pi^{-1}(W)$. We work with (ii). Let $W \subset S$ be an irreducible algebraic subvariety and $Y \subset \pi^{-1}W$ an irreducible algebraic subvariety of X^+ , maximal for these properties. We want to show that $\pi(Y)$ is weakly special. The main intermediate step is the following:

Proposition 8.1. There exists a connected \mathbb{Q} -algebraic subgroup \mathbf{H}_Y of \mathbf{G} , of positive dimension, such that $\mathbf{H}_Y(\mathbb{R})^+ \subset \operatorname{Stab}_{\mathbf{G}(\mathbb{R})^+}(Y)$.

Let us show how to deduce Theorem 4.25 from Proposition 8.1. The arguments are close to the ones used in the proof of Theorem 4.7.

Let \mathbf{H}_Y be the largest connected \mathbb{Q} -algebraic subgroup of \mathbf{G} such that $\mathbf{H}_Y(\mathbb{R})^+ \subset \operatorname{Stab}_{\mathbf{G}(\mathbb{R})^+}(Y)$. By Proposition 8.1 the group \mathbf{H}_Y is positive dimensional.

Let $W' \subset S$ be the Zariski-closure of $\pi(Y)$. Replacing W by W' we can assume that $\pi(Y)$ is Zariski-dense in W. One also can assume that W is Hodge generic, replacing S by the smallest special subvariety of S containing W. In this situation it follows that $\pi(Y)$ is also Hodge-generic in the sense that $\pi(Y)$ is not contained in any strict special subvariety S' of S. Otherwise $\pi(Y) \subset S' \cap W \subsetneq W$ contradicting the Zariski-density of $\pi(Y)$ in W.

Lemma 8.2. Let \tilde{W} be an irreducible component of $\pi^{-1}(W)$ containing Y. Then $\mathbf{H}_Y(\mathbb{Q})$ stabilizes \tilde{W} .

Proof. Let $h \in \mathbf{H}_Y(\mathbb{Q})$. As $Y \subset \tilde{W} \cap h\tilde{W}$ is irreducible algebraic there exists an irreducible component Z of $\tilde{W} \cap h\tilde{W}$ containing Y. Notice that $\pi(Z)$ is an irreducible component of $W \cap T_h(W)$ containing $\pi(Y)$. As $\pi(Y)$ is Zariski-dense in W it follows that $\pi(Z) = W$. Hence $\tilde{W} = h\tilde{W}$.

Without loss of generality we can assume that \mathbf{G} is semi-simple of adjoint type. Indeed consider the morphism of Shimura data $\psi: (\mathbf{G}, X) \longrightarrow (\mathbf{G}^{\mathrm{ad}}, X^{\mathrm{ad}})$. Let $K^{\mathrm{ad}} \subset \mathbf{G}^{\mathrm{ad}}(\mathbb{A}_{\mathrm{f}})$ be a compact open subgroup containing the image of K. We thus have a morphism of Shimura varieties $\psi: \mathrm{Sh}_K(\mathbf{G}, X) \longrightarrow \mathrm{Sh}_{K^{\mathrm{ad}}}(\mathbf{G}^{\mathrm{ad}}, K^{\mathrm{ad}})$ and the conjectures for W and $\psi(W)$ are equivalent.

For simplicity let us first assume that G is \mathbb{Q} -simple.

We choose a Hodge-generic point $z \in W^{\mathrm{sm}}$ and a point $\tilde{z} \in \tilde{W}$ above z. Let $\rho : \pi_1(W^{\mathrm{sm}}, z) \longrightarrow \Gamma \subset \mathbf{GL}(V_{\mathbb{Z}})$ be the associated monodromy representation with image $\Gamma_W := \rho(\pi_1(W^{\mathrm{sm}}, z)) \subset \Gamma$. By Galois theory Γ_W is the subgroup of Γ stabilizing \tilde{W} . In particular the group Γ_W contains

$$\mathbf{H}_Y(\mathbb{Z}) := \mathbf{H}_Y(\mathbb{Q}) \cap \Gamma = \mathbf{H}_Y(\mathbb{Q}) \cap \mathbf{G}(\mathbb{Z}).$$

Deligne's Theorem 4.8 then states that the Zariski-closure $\overline{\Gamma_W}$ of Γ_W is normal in \mathbf{G} . As we assumed that \mathbf{G} is simple, it follows that $\overline{\Gamma_V} = \mathbf{G}$.

Lemma 8.3. The group Γ_W normalizes \mathbf{H}_Y .

Proof. Let $\gamma \in \Gamma_W$. Thus $\gamma \mathbf{H}_Y \gamma^{-1} \cdot \tilde{W} = \tilde{W}$. Hence

$$Y' := \gamma \mathbf{H}_Y \gamma^{-1} \cdot Y \subset \tilde{W}$$
.

But Y' is semi-algebraic and contains Y. In this situation Y' is contained in an irreducible algebraic subvariety of X^+ contained in \tilde{W} and maximal for these properties. By our maximality assumption on Y it follows that Y = Y'. Hence $\gamma \mathbf{H}_Y \gamma^{-1}$ fixes Y and it follows that $\gamma \mathbf{H}_Y \gamma^{-1} = \mathbf{H}_Y$.

We finish the proof of Theorem 4.25 in this case by noticing that the normaliser of \mathbf{H}_Y is algebraic and contains Γ_W . Hence it contains $\overline{\Gamma}_W = \mathbf{G}$. As we supposed that \mathbf{G} is simple if follows that $\mathbf{G} = \mathbf{H}_Y$. Hence \mathbf{G} stabilizes \tilde{W} and Y. Finally $Y = \tilde{W} = X^+$ and $\pi(Y) = W = S$.

In general the adjoint group \mathbf{G} is a product of simple factors. One obtains a decomposition $(\mathbf{G}, X) = (\mathbf{G}_1, X_1) \times (\mathbf{G}_2, X_2)$ with \mathbf{G}_1 the Zariski-closure of the monodromy Γ_W . The same kind of arguments as in the simple case then show that

$$\pi(Y) = W = \pi(X_1^+ \times \{x_2\})$$

for some point $x_2 \in X_2$.

8.3. O-minimal arguments and hyperbolic geometry. To prove Proposition 8.1 we introduce the set

$$\Theta(Y) := \{ g \in \mathbf{G}(\mathbb{R}) : \dim(gY \cap \pi^{-1}V \cap \mathcal{F}) = \dim(Y) \} ,$$

where \mathcal{F} is a fundamental set for the action of Γ on X^+ as in Theorem 6.2.

Theorem 6.2 implies that $\Theta(Y)$ is definable in $\mathbb{R}_{an,exp}$. This relies on the fact that the dimension function is a well-defined definable function in any o-minimal theory [vdD98].

The inclusion $g \cdot Y \subset \pi^{-1}(W)$ holds for any $g \in \Theta(Y)$. This follows from the inclusion $g \cdot Y \cap \mathcal{F} \subset \pi^{-1}(W)$ and analytic continuation.

Lemma 8.4.

$$\Theta(Y) \cap \Gamma = \{ \gamma \in \Gamma, \quad \gamma^{-1} \mathcal{F} \cap Y \neq \emptyset \} .$$

Moreover for any $\gamma \in \Theta(Y) \cap \Gamma$ the translate $\gamma \cdot Y$ is irreducible algebraic in $\pi^{-1}(W)$, maximal for these properties.

Proof. The Γ -invariance of $\pi^{-1}(V)$ implies:

$$\Theta(Y) \cap \Gamma = \{ \gamma \in \Gamma, \quad \dim(\gamma Y \cap \pi^{-1} V \cap \mathcal{F}) = \dim(Y) \}$$
$$\{ \gamma \in \Gamma, \quad \dim(Y \cap \gamma^{-1} \mathcal{F}) = \dim(Y) \} .$$

As \mathcal{F} is open in X^+ the conditions $\dim(Y \cap \gamma^{-1}\mathcal{F}) = \dim(Y)$ and $\gamma^{-1}\mathcal{F} \cap Y \neq \emptyset$ are the same. The first part of the lemma follows. The second part follows from the inclusion $\gamma \cdot Y \subset \pi^{-1}(V)$ obtained by analytic continuation as above and the maximality of Y among the irreducible algebraic subvarieties of X^+ contained in $\pi^{-1}(W)$.

The heart of the proof of Proposition 8.1 is the following statement. Let H be the height function H on $\mathbf{G}(\overline{\mathbb{Q}})$ deduced from the canonical height function on $\mathrm{End}(V_{\overline{\mathbb{Q}}}) \simeq \overline{\mathbb{Q}}^{n^2}$ and the embedding $\mathbf{G}(\overline{\mathbb{Q}}) \subset \mathbf{GL}(V_{\overline{\mathbb{Q}}}) \subset \mathrm{End}(V_{\overline{\mathbb{Q}}})$. For every positive real number T we define

$$N_Y(T) := \{ \gamma \in \Gamma, \quad Y \cap \gamma^{-1} \mathcal{F} \neq \emptyset, \quad H(\gamma) \le T \} .$$

Theorem 8.5. There exists a positive real number a such that for T large enough

$$N_Y(T) \ge T^a$$
.

Indications on the proof of Theorem 8.5 will be given in the next section. For now let us show how it implies Proposition 8.1.

First notice that if B is a semi-algebraic bloc of $\Theta(Y)$ containing an element $\gamma \in \Theta(Y) \cap \Gamma$ then

$$B \subset \gamma \cdot \operatorname{Stab}_G(Y)$$
.

Indeed if U_{γ} is an open semi-algebraic subset of B containing γ then $U_{\gamma} \cdot Y$ is semi-algebraic contained in $\pi^{-1}(W)$ and contains the maximal algebraic Y of $\pi^{-1}(Y)$. Hence $U_{\gamma} \cdot Y = \gamma Y$. For $b \in B$ one can construct a connected semi-algebraic set $U(\gamma, b)$ of B containing γ and b. The same argument shows that

$$\gamma \cdot Y = b \cdot Y = B \cdot Y .$$

Applying the bloc version Theorem 5.13 of Pila-Wilkie's counting theorem, we obtain positive real numbers b_1 and b_2 such that for T sufficiently large, there exists a bloc B in $\Theta(Y)$ such that

$$|\{\gamma \in B \cap \Gamma, H(\gamma) \le T^{b_1}\}| \ge T^{b_2}$$
.

If we fix $\gamma_0 \in B \cap \Gamma$ the previous discussion shows that the subset $\gamma_0^{-1} \cdot (B \cap \Gamma) \subset \operatorname{Stab}_G(Y)$ contains at least T^{b_2} elements. It follows that $\operatorname{Stab}_G(Y) \cap \Gamma$ is infinite. Hence the algebraic subgroup of \mathbf{G} generated by $\operatorname{Stab}_G(Y) \cap \Gamma$ is positive dimensional. This finishes the proof that Theorem 8.5 implies Proposition 8.1.

8.4. An algebraic curve of X^+ meets many fundamental domains. Theorem 8.5, which is the technical heart of the proof of Theorem 4.25, is a statement in hyperbolic geometry. We have to show that an irreducible algebraic subvariety Y of X^+ cuts "many" Γ -translates of \mathcal{F} . Hence we can assume that Y is the intersection C of an irreducible algebraic curve \hat{C} of \hat{X} with X^+ .

The following comparison between the norm and the distance on X^+ on the one hand, the norm and the height on the other hand is a nice exercise on locally symmetric spaces:

Lemma 8.6. (i) For any $g \in \mathbf{G}(\mathbb{R})$ the following inequality holds:

(8.1)
$$\log ||g||_{\infty} \le d(g \cdot x_0, x_0) .$$

(ii) There exists a positive number B and a positive integer N such that:

(8.2)
$$\forall \gamma \in \mathbf{G}(\mathbb{Z}), \quad \forall u \in \gamma \mathcal{F}, \qquad H(\gamma) \leq B \cdot ||u||_{\infty}^{n}.$$

We also have at our disposal a lower bound for the volume of complex-analytic subvariety of X^+ due to Hwang and To [HwTo02]. Let us denote by Vol_C the area on C for the restriction of the metric g_X to C. For a positive real number R we denote by $B(x_0, R)$ the geodesic ball of X^+ of center x_0 and radius R.

Theorem 8.7. Let C be a complex analytic curve in X^+ . For any point $x_0 \in C$ there exist positive constants a, b such that for any positive real number R one has:

(8.3)
$$\operatorname{Vol}_{C}(C \cap B(x_{0}, R)) \geq a \exp(b \cdot R) .$$

The key lemma for the proof of Theorem 8.5 is then the following upper-bound for the volume of an algebraic curve (the proof uses the geometry of toroidal compactifications):

Lemma 8.8. There exists a constant A > 0 such that for any algebraic curve $C \subset X^+$ of degree d we have the bound

(8.4)
$$\operatorname{Vol}_{C}(C \cap \mathcal{F}) \leq A \cdot d .$$

With all these ingredients we show Theorem 8.5 as follows. Let T be a positive real number. Let us define

(8.5)
$$C(T) := \{ u \in C \text{ and } ||u||_{\infty} \leq T \}$$
$$= \bigcup_{\substack{\gamma \in \Gamma \\ \gamma \mathcal{F} \cap C \neq \emptyset}} \{ u \in \gamma \mathcal{F} \cap C \text{ and } ||u||_{\infty} \leq T \}$$

It follows from the (8.2) that

(8.6)
$$C(T) \subset \bigcup_{\substack{\gamma \in \Gamma, \ \gamma \mathcal{F} \cap C \neq \emptyset \\ H(\gamma) \leq B \cdot T^N}} \{ u \in \gamma \mathcal{F} \cap C \} .$$

Taking volumes:

(8.7)
$$\operatorname{Vol}_{C}(C(T)) \leq \sum_{\substack{\gamma \in \Gamma, \ \gamma \mathcal{F} \cap C \neq \emptyset \\ H(\gamma) \leq B \cdot T^{N}}} \operatorname{Vol}_{C}(\mathcal{F} \cap \gamma^{-1}C) ,$$

hence

(8.8)
$$\operatorname{Vol}_{C}(C(T)) \leq \sum_{\substack{\gamma \in \Gamma, \ \gamma \mathcal{F} \cap C \neq \emptyset \\ H(\gamma) \leq B \cdot T^{N}}} \operatorname{Vol}_{C}(\mathcal{F} \cap \gamma^{-1}C) .$$

Notice that all the curves $\gamma^{-1}C$, $\gamma \in \mathbf{G}(\mathbb{Z})$, have the same degree as algebraic curves. Hence it follows from (8.4) that

(8.9)
$$\operatorname{Vol}_{C}(C(T)) \leq (A \cdot d) \cdot N_{C}(B \cdot T^{N}) .$$

Observe that Part (i) of Lemma 8.6 implies that $C \cap B(x_0, \log T) \subset C(T)$. Thus:

(8.10)
$$\operatorname{Vol}_{C}(C \cap B(x_{0}, \log T)) \subset \operatorname{Vol}_{C}C(T) .$$

Using inequality (8.9) and Theorem 8.7 it follows that

$$aT^b \le A \cdot d \cdot N_C(B \cdot T^N)$$
.

This finishes the proof of Theorem 8.5.

9. Lower bounds for Galois orbits of CM-points

9.1. Class groups for tori and reciprocity morphisms.

9.1.1. Class groups for tori. Let **M** be an algebraic torus over \mathbb{Q} . We denote by K_M^m the unique maximal compact subgroup of $\mathbf{M}(\mathbb{A}_f)$.

Definition 9.1. The absolute class group of M is the finite group

$$h_{\mathbf{M}} := \mathbf{M}(\mathbb{Q}) \backslash \mathbf{M}(\mathbb{A}_{\mathrm{f}}) / K_{\mathbf{M}}^{m}$$
.

If $K_{\mathbf{M}} \subset \mathbf{M}(\mathbb{A}_{\mathrm{f}})$ is an arbitrary compact open subgroup we define the associated relative class group as the finite group

$$h_{\mathbf{M},K_{\mathbf{M}}} := \mathbf{M}(\mathbb{Q}) \backslash \mathbf{M}(\mathbb{A}_{\mathrm{f}}) / K_{\mathbf{M}}$$
,

so that $h_{\mathbf{M}} = h_{\mathbf{M}, K_{\mathbf{M}}^m}$.

Notice that if F is a number field and $\mathbb{R}_F := \operatorname{Res}_{F/\mathbb{O}} \mathbb{G}_{m,F}$ then $h_{\mathbb{R}_F}$ is equal to the classical class group h_F of the ring of integers \mathcal{O}_F of F.

9.1.2. Reciprocity morphisms. The notations are those of Section 2. Let x = [x, 1]be a CM-point of S. The Mumford-Tate group \mathbf{MT}_x is a \mathbb{Q} -torus \mathbf{T} and $(\mathbf{T}, \{x\})$ is a Shimura sub-datum of (\mathbf{G}, X) . Let $K_{\mathbf{T}} := K \cap \mathbf{T}(\mathbb{A}_{\mathrm{f}})$. Then

$$\operatorname{Sh}_{K_{\mathbf{T}}}(\mathbf{T}, \{x\}) = \mathbf{T}(\mathbb{Q}) \setminus (\{x\} \times \mathbf{T}(\mathbb{A}_{\mathrm{f}})/K_{\mathbf{T}}) \subset \operatorname{Sh}_{K}(\mathbf{G}, X)$$

is a zero-dimensional subvariety, of cardinality h_{T,K_T} , defined over the reflex field E := $E(\mathbf{T}, \{x\}) \text{ of } (\mathbf{T}, \{x\}).$

The theory of Complex Multiplication gives a surjective morphism, called the reciprocity morphism

$$r := r(\mathbf{T}, \{x\}) : \mathbb{R}_E \longrightarrow \mathbf{T}$$
.

9.2. Faltings height. Let K be a number field and A_K an Abelian variety over K of dimension g. Let $p: \mathbf{A} \longrightarrow \operatorname{Spec}(\mathcal{O}_K)$ be its Néron model and $\epsilon: \operatorname{Spec}(\mathcal{O}_K) \longrightarrow \mathbf{A}$ its unit section. We denote by $\omega_{\mathbf{A}_K} := \epsilon^* \Omega^g_{\mathbf{A}/\operatorname{Spec}\mathcal{O}_K}$. Every field embedding $\sigma : K \longrightarrow \mathbb{C}$ defines a Hermitian metric on

$$\omega_{\mathbf{A}_K,\sigma} := H^0(A_{\sigma}(\mathbb{C}), \Omega^g_{A_{\sigma}(\mathbb{C})})$$

given on any section $\alpha \in H^0(A_{\sigma}(\mathbb{C}), \Omega^g_{A_{\sigma}(\mathbb{C})})$ by

$$||\sigma||_{\sigma} := \left| \frac{1}{(2\pi)^g} \alpha \wedge \overline{\alpha} \right|.$$

We denote $\overline{\omega_{\mathbf{A}_K}}$ the element $(\omega_{\mathbf{A}_K}, ||\cdot||_{\sigma}) \in \operatorname{Pic}(\mathcal{O}_K)$.

The Faltings height of A is defined as

$$h_F(A) := \deg_{A_F}(\overline{\omega_{A_K}})$$
.

If A has semi-stable reduction over K the Faltings height $h_F(A)$ does not change under base change to a finite extension of K. If A has good reduction over K there exists a finite extension L of K such that $\omega_{\mathbf{A}_L} \simeq \mathcal{O}_L$. Choosing a Neron differential $\omega \in \omega_{\mathbf{A}_L}$, one then obtains

$$h_F(A) = -\frac{1}{[L:\mathbb{Q}]} \sum_{\sigma: L \to \mathbb{Q}} \log ||\omega||_{\sigma} .$$

9.3. Lower bounds for Galois orbits. In 2001, motivated by applications to the André-Oort conjecture, Edixhoven formulated a conjecture that Galois orbits of CM abelian varieties of fixed dimension should grow as a uniform power of the discriminant of the center of the Endomorphism ring of the abelian varieties in question. He stated his conjecture precisely as Problem 14 in [EMO]. In private communications with the third author in 2000, Edixhoven expressed the view that his conjecture should follow easily from the Colmez conjecture on Faltings height of CM abelian varieties. Reference to Colmez' paper appears explicitly in the statement of his Problem 14 of [EMO]. It appears that before Tsimerman's announcement, the required inequality relating Faltings height to the discriminant of the centre of the Endomorphism algebra has been known to experts on transcendence theory and diophantine approximation. In his 2005 thesis Eric Villani (see [V]) explicitly works out the required inequality for abelian varieties corresponding to special points of Hilbert modular varieties as a subproduct of his main result - see 'Remarque' on page 18 of [V]. In [GR14], the authors develop all the necessary tools for the deduction and in [R], Proposition 2.10, Gaél Rémond explicitly states the deduction of the required inequality from the results of [GR14]. Until the announcements of proofs of the Colmez conjecture on average in 2014 (see [AGHM] and [YuZh]), there was very little progress on Colmez conjecture. Upon the announcement, Tsimerman immediately presented an argument for the deduction of lower bounds of Galois orbits from it, which, as explained above, has been essentially well-known to the experts. In this section we give the argument as presented by Tsimerman.

The Faltings height can be interpreted as a height on the set $\mathcal{A}_g(\mathbb{Q})$ of algebraic points of \mathcal{A}_g . If $x \in \mathcal{A}_g(\mathbb{Q})$ parameterizes the Abelian variety A_x one define $h_F(x) = h_F(A_x)$. This function satisfies the Northcott property: given d and T positive real integers the set

$$N_{d,T}(\mathcal{A}_q) := \{ x \in \mathcal{A}_q(\overline{\mathbb{Q}}), [\mathbb{Q}(x) : \mathbb{Q}] \le d \text{ and } h_F(x) \le T \}$$

is finite. If the Falting's height h_F were uniformly bounded on CM-point of \mathcal{A}_g we would directly obtain that the fields of definition of these points have a degree tending to infinity. This type of argument is used in the proof of the Manin-Mumford conjecture to obtain a lower bound or Galois-orbits of torsion points of an Abelian variety, as these are the points of canonical height zero. For \mathcal{A}_g it is not true that the Faltings height is uniformly bounded but a direct consequence of the Colmez conjecture on average, (which we describe in the next section) is the following:

Theorem 9.2. Let g be a positive integer and ϵ a positive real number. There exists a positive real number $c_1 = c_1(g, \epsilon)$ with the following property. Let E be a CM-field of degree 2g with discriminant d_E . Let A be a g-dimensional Abelian variety with complex multiplication by the ring of integers \mathcal{O}_E of E. Then

$$h_F(A) \leq c_1 |d_E|^{\epsilon}$$
.

The main result of Tsimerman in [Tsi] is the following corollary for the size of Galois orbits of CM-points in \mathcal{A}_g :

Theorem 9.3. (Tsimerman) Let g be a positive integer. There exist positive real numbers $\alpha = \alpha(g)$ and $c_2 = c_2(g)$ with the following property. Let E be a CM-field of degree 2g with discriminant d_E . Let $x \in \mathcal{A}_g(\overline{\mathbb{Q}})$ be a CM-point parameterizing an Abelian

variety with complex multiplication by \mathcal{O}_E . Then

$$[\mathbb{Q}(x):\mathbb{Q}] \ge c_2 \cdot |d_E|^{\alpha}$$
.

The proof of Theorem 9.3 relies on a deep theorem of Masser and Wüstholz [MaWü95] (which is also the crux of an alternative proof of Mordell's conjecture):

Theorem 9.4. (Masser-Wüstholz) Let g be a positive integer. There exist positive real numbers $\beta = \beta(g)$ and $c_3 = c_3(g)$ with the following property. Let A and B be two Abelian varieties defined over a number field k. We suppose that A and B are $\overline{\mathbb{Q}}$ -isogenous. Then there exists a $\overline{\mathbb{Q}}$ -isogeny from A to B of degree N with

$$N \leq c_2 \max(h_F(A), [k:\mathbb{Q}])^{\beta}$$
.

Tsimerman's argument is the following. Let Σ be the locus in \mathcal{A}_g of Abelian varieties with complex multiplication by \mathcal{O}_E and fixed CM-type Φ . For all $x, y \in \Sigma$ the Abelian varieties A_x and A_y are $\overline{\mathbb{Q}}$ -isogenous. On the other hand the cardinal of Σ is the cardinal of the class group of \mathcal{O}_E . As E is CM the class formula gives $|\Sigma| \gg d_E^{\gamma}$ for an absolute constant $\gamma > 0$ for d_E sufficiently large.

Let us fix $x_0 \in \Sigma$. Let N be a positive integer. There exists $\delta > 0$ such that the number of $\overline{\mathbb{Q}}$ -isogenies with source A_{x_0} of degree at most N is bounded above by N^{δ} for N sufficiently large. Let η be a positive real number such that $\eta < \frac{\gamma}{\delta}$. Taking $N = d_E^{\eta}$ and d_E large enough it follows that there exists $x \in \Sigma$ such that the minimal degree $d_{\min}(A_{x_0}, A_x)$ of a $\overline{\mathbb{Q}}$ -isogeny from A_{x_0} to A_x satisfies

$$d_{\min}(A_{x_0}, A_x) > d_E^{\eta} .$$

By the Masser-Wüstholz Theorem 9.4 and the upper-bound on the Faltings height given by Theorem 9.2 it follows that:

$$d_E^{\eta} \le c_3 \max(h_F(A_{x_0}), [k:\mathbb{Q}]^{\beta}) \le c_3 \max(c_1 d_E^{\epsilon}, k:\mathbb{Q}]^{\beta}) .$$

If we choose $\epsilon < \eta$ and d_E sufficiently large one obtains

$$[k:\mathbb{Q}] \ge d_E^{\frac{\eta}{\beta}} \ .$$

9.4. Colmez conjecture. The reference for this section is [Col93]. Let A be a simple Abelian variety over \mathbb{C} , with complex multiplication and of dimension g. The field $E := \operatorname{End}(A) \otimes \mathbb{Q}$ is CM with $[E : \mathbb{Q}] = 2g$. We suppose moreover that $\operatorname{End} A = \mathcal{O}_E$. Let $\Phi \subset \operatorname{Hom}(E, \mathbb{C})$ be the CM type of A. Hence

$$\operatorname{Lie}(A) = \bigoplus_{\sigma \in \Phi} \operatorname{Lie}(A)_{\sigma} ,$$

where $\text{Lie}(A)_{\sigma}$ is the subspace of Lie(A) on which E acts through σ .

Let K be a number field on which A is defined and has good reduction. Colmez shows that the height $h_F(A)$ depends on (E, Φ) only and conjectures a closed formula for $h_F(A)$. We will write $h_F(A) = h_F(\Phi)$ in the sequel.

Let F be a field extension of \mathbb{Q} . We denote by \mathcal{G}_F the Galois group $\operatorname{Gal}(\overline{\mathbb{Q}}/F)$ and by $c \in \mathcal{G}_{\mathbb{Q}}$ the complex conjugation. Let $C(\mathcal{G}_{\mathbb{Q}}, \mathbb{C})$ be the complex vector space of locally constant complex functions on $\mathcal{G}_{\mathbb{Q}}$ and $C^0(\mathcal{G}_{\mathbb{Q}}, \mathbb{C})$ its subspace of central ones.

Let $\mathbb{Q}^{\mathrm{CM}} \subset \overline{\mathbb{Q}}$ be the extension of \mathbb{Q} generated by CM-fields. This is a Galois extension of \mathbb{Q} . We denote by $\mathrm{CM}^0(\mathcal{G}_{\mathbb{Q}},\mathbb{C}) \subset C^0(\mathcal{G}_{\mathbb{Q}},\mathbb{C})$ the subspace of functions f such that $f(\sigma)$ depends only on the $\mathcal{G}_{\mathbb{Q}^{\mathrm{CM}}}$ -conjugacy class of σ and such that $f(\sigma)+f(c\sigma)$ is independent of σ .

We define a Hermitian scalar product <,> on $C(\mathcal{G}_{\mathbb{Q}},\mathbb{C})$ by:

$$\forall \Theta_1, \Theta_2 \in C(\mathcal{G}_{\mathbb{Q}}, \mathbb{C}), \quad <\Theta_1, \Theta_2 > := \frac{1}{|\mathcal{G}_{\mathbb{Q}}/\mathcal{G}_{\mathcal{F}}|} \sum_{g \in \mathcal{G}_{\mathbb{Q}}/\mathcal{G}_{\mathcal{F}}} \Theta_1(g) \overline{\Theta_2(g)} ,$$

where F is any finite extension of \mathbb{Q} such that Θ_1 and Θ_2 depend only on conjugacy classes modulo \mathcal{G}_F .

The set Art of Artin characters (i.e. characters of continuous finite dimensional complex representations of $\mathcal{G}_{\mathbb{Q}}$) is an orthonormal basis of $C^0(\mathcal{G}_{\mathbb{Q}}, \mathbb{C})$. Given any Artin character χ , we denote by $L(\chi, s)$ its L-function. One also checks that the set of Artin characters whose L-function does not vanish at 0 form an orthonormal basis of $\mathrm{CM}^0(\mathcal{G}_{\mathbb{Q}}, \mathbb{C})$.

For $\Theta \in C(\mathcal{G}_{\mathbb{Q}}, \mathbb{C})$ we denote by Θ^0 its orthonormal projection

$$\Theta^0 = \sum_{\chi \in Art} <\Theta, \chi > \chi$$

on $C^0(\mathcal{G}_{\mathbb{O}}, \mathbb{C})$.

We also denote by $Z(\chi, s)$ the logarithmic derivative $L'(\chi, s)/L(\chi, s)$ and by $\mu_{Art}(\chi)$ the logarithm log f_{χ} of the Artin conductor f_{χ} of χ . These functions admit local decompositions

$$\begin{split} \mu_{\text{Art}} &= \sum_{p \text{ premier}\}} \mu_{\text{Art},p} \log p \;\;, \\ \forall \operatorname{Re}(s) &> 1, \quad Z(\chi,s) = -\sum_{p \text{ premier}} Z_p(\chi,s) \log p \;\;. \end{split}$$

For any prime p, the local factor $Z_p(\chi, s)$ lies in $\mathbb{Q}(p^{-s})$. The function $Z(\chi, s)$ admits an holomorphic extension to \mathbb{C} and a functional equation.

Given a CM-type (E, Φ) we define the function $A_{\Phi} \in C(\mathcal{G}_{\mathbb{Q}}, \mathbb{C})$ by

$$A_{\Phi}(g) = \frac{|\Phi \cap g\Phi|}{[E:\mathbb{Q}]}$$

and denote by A_{Φ}^0 its projection on $C^0(\mathcal{G}_{\mathbb{Q}},\mathbb{C})$. One checks that $A_{\Phi}^0 \in \mathrm{CM}^0(\mathcal{G}_{\mathbb{Q}},\mathbb{C})$. Colmez conjecture is the following:

Conjecture 9.5. Let A be a complex Abelian variety of CM-type (E, Φ) . Then:

$$h_F(A) = Z(A_{\Phi}^0) - \frac{1}{2}\mu_{\text{Art}}(A_{\Phi}^0) = -\sum_{\chi \in \text{Art}} \langle A_{\Phi}, \chi \rangle \left(\frac{L'(0, \chi)}{L(0, \chi)} + \frac{1}{2}\mu_{\text{Art}}(\chi)\right) .$$

Let F be the totally real subfield of E, F its discriminant and $d_{E/F} := N_{E/F}d_E$ the relative discriminant of E over F. Let $\chi_{E/F}$ be the associated quadratic character of F. As noticed by Colmez, Conjecture 9.5 simplifies if we average on the 2^g possible CM-types of E. It is this result which is proved by completely different method by Andreatta-Goren-Howard-Madapusi Pera [AGHM] and Yuan-Zhang [YuZh] and which implies Theorem 9.2:

Theorem 9.6. (Colmez conjecture on average)

$$\frac{1}{2^g} \sum_{\Phi} h_F(\Phi) = -\frac{1}{2} \frac{L'(\chi_{E/F}, 0)}{L(\chi_{E/F}, 0)} - \frac{1}{4} \log |d_{E/F} d_F| .$$

10. Further Developments: the André-Pink conjecture.

In this section we briefly present a conjecture which is a special case of the Zilber-Pink conjecture and which is in some sense a more natural analog of the Manin-Mumord conjecture in the context of (mixed) Shimura varieties. This conjecture is now usually referred to as the André-Pink conjecture and was stated explicitly by Pink in [Pink05]. Pink also obtained a result on this conjecture under certain quite strong assumptions. We will not touch upon Pink's work in this paper and refer to Pink's (excellent) exposition in [Pink05]. The general statement of the André-Pink conjecture is as follows ([An89], Problem 3 and [Pink05], Conjecture 1.6):

Conjecture 10.1 (André-Pink). Let S be a mixed Shimura variety over \mathbb{C} and $\Lambda \subset S$ be the generalised Hecke orbit of a point s of S. Let Z be a subvariety of S such that $Z \cap \Lambda$ is Zariski dense in Z. Then Z is a weakly special subvariety.

The André-Pink conjecture is still open in general. However Martin Orr (see [Orr15]) has obtained a fairly general result, using the techniques explained in this text. The primary aim of this section is to explain Orr's result and give an idea of its proof.

Instead of looking at the Zariski closure of a set of special points, one looks at the Zariski closure of a subset of a (generalised) Hecke orbit in a (mixed) Shimura variety. The expectation is that components of this Zariski closure are weakly special. In the case of \mathcal{A}_g the conjecture becomes the following.

Conjecture 10.2. Let Λ be the isogeny class of a point $s \in \mathcal{A}_g(\mathbb{C})$. Let Z be an irreducible closed subvariety of \mathcal{A}_g such that $Z \cap \Lambda$ is Zariski dense in Z. Then Z is a weakly special subvariety of \mathcal{A}_g .

In the case where s is Galois-generic, Pink in [Pink05], proves that Conjecture 10.2 follows from results of Clozel, Oh and Ullmo ([CUO01]) on equidistribution of Hecke orbits. However, it would be a relatively rare occurence that s is Hodge generic. In the case where s is a special point, Conjecture 10.2 is a special case of the André-Oort conjecture, known for \mathcal{A}_q . In [Orr15], M. Orr proves the following.

Theorem 10.3 (Orr). Let s and Λ and Z be as in 10. Then:

(1) There exists a special subvariety $S \subset \mathcal{A}_g$, isomorphic to a product $S_1 \times S_2$ of Shimura varieties, such that $\dim(S_1) > 0$ and

$$Z = S_1 \times Z' \subset S$$

where Z' is a closed subvariety of S_2 .

(2) If Z is a curve, then Z is weakly special.

It is of course obvious that (2) follows from (1).

The strategy of Orr's proof is again a combination of lower bounds for Galois orbits with Pila-Wilkie theorem (the blocks version - Theorem 5.13), Ax-Lindemann (Theorem 4.25) and Ullmo's Theorem 7.2. Note that elaboration of suitable lower bounds for the Galois orbits makes essential use of the Masser-Wüstholz theorem (Theorem 9.4).

Let s be a point of $\mathcal{A}_g(\mathbb{C})$, Λ its isogeny class and Z an irreducible subvariety of \mathcal{A}_g such that $Z \cap \Lambda$ is Zariski dense.

We call \mathbf{H}_g the Siegel upper half space and $\pi\colon \mathbf{H}_g \longrightarrow \mathcal{A}_g$ the uniformization map. We also call \mathcal{F}_g the classical Siegel fundamental domain. Let

$$\widetilde{Z} = \mathcal{F}_q \cap \pi^{-1} Z$$
 and $\widetilde{\Lambda} = \mathcal{F}_q \cap \pi^{-1} \Lambda$.

Given a point s of $\mathcal{A}_g(\mathbb{C})$, we let A_s be the abelian variety associated to s. We define the complexity of t in Λ as the minimum degree of isogeny between A_s and A_t .

Similarly, we define the complexity of a point t in Λ .

The height of a matrix in $M_n(\mathbb{Q})$ is defined as the maximum of heights of its entries. Orr proves the following:

Proposition 10.4 ([Orr15], Proposition 3.2). Let Z be a subvariety of A_g and \widetilde{s} a point in \mathcal{F}_g . Let $\epsilon > 0$. There exists an $C = c(Z, \widetilde{s}, \epsilon)$ such that for every $n \geq 1$, there is a collection of at most cn^{ϵ} semi-algebraic blocks $W_i \subset \widetilde{W}$ such that all points of $\widetilde{Z} \cap \widetilde{\Lambda}$ of complexity $\leq n$ are contained in $\bigcup_i W_i$.

The idea of the proof is to construct a certain definable subset Y of $\mathbf{GL}_{2g}(\mathbb{R})$, show that it contains 'a lot' of points of $\mathbf{GL}_{2g}(\mathbb{Q})$ up to height n and then apply Pila-Wilkie theorem (block version) to it.

The crucial lemma is the following which is of independent interest.

Lemma 10.5 ([Orr15], Lemma 3.3). There exist constant c, k depending only on g and \widetilde{s} such that: for any $\widetilde{t} \in \widetilde{Z} \cap \widetilde{\Lambda}$ of complexity n, there is a rational matrix $\gamma \in Y$ such that $\gamma \widetilde{s} = \widetilde{t}$ and the height of γ is at most cn^k .

On the other hand, Masser-Wustholz theorem gives a polynomial (in the complexity) lower bound on the size of the Galois orbits of the points of Λ .

This implies, via Pila-Wilkie theorem and Ax-Lindemann, that positive dimensional weakly special subvarieties are dense in Z. Ullmo's Theorem 7.2 then implies the conclusion of Theorem 10.3.

References

[An89] Y. André, G-functions and geometry, Aspects of Mathematics E13 (1989)

[An92] Y. André, Mumford-Tate groups of mixed Hodge structures and the theorem of the fixed part, Compositio Math. 82 (1992) 1-24

[An98] Y. André, Finitude des couples d'invariants modulaires singuliers sur une courbe algébrique plane non modulaire, J. Reine Angew. Math. 505 (1998), 203-208

[AGHM] F. Andreatta, E. Goren, B. Howard, K. Madapusi-Pera, Faltings heights of Abelian varieties with complex multiplication, preprint 2015

[AMRT75] A. Ash, D. Mumford, M. Rapoport, Y. Tai, Smooth Compactification of Locally Symmetric varieties, Lie Groups: History Frontiers and applications vol. 4. Math Sci Press. (1975).

[Ax71] J. Ax, On Schanuel's conjecture, Annals of Math. 93 (1971), 1-24

[BB66] W.L. Baily, A. Borel, Compactification of arithmetic quotients of bounded symmetric domains, Annals of Math., 84 (1966), 442-528

- [Bor69] A. Borel, *Introduction aux groupes arithmétiques*, Publications de l'Institut de Mathématique de l'Université de Strasbourg, XV. Actualités Scientifiques et Industrielles, No. 1341 Hermann, Paris (1969)
- [CDK95] É. Cattani, P. Deligne, A. Kaplan, On the locus of Hodge classes. J. of AMS, 8 (1995), 483-506
 [CloUl05] L. Clozel, E. Ullmo, Equidistribution de sous-variétés spéciales, Annals of Math. 161 (2005), 1571-1588.
- [CUO01] L. Clozel, H. Oh, E. Ullmo Hecke operators and equidistribution of Hecke points, Invent. Math. 144, (2001), p. 327-351
- [Co96] P. Cohen, Humbert surfaces and transcendence properties of automorphic functions, Rocky Mountain J. Math. 26 (1996), no.3, 987-1001
- [Col93] P. Colmez, Périodes des variétés abéliennes à multiplication complexe, Annals of Math. 138 (1993), no.3, 625-683
- [Col98] P. Colmez, Sur la hauteur de Faltings des variétés abéliennes à multiplication complexe, Compositio Math. 111 (1998), no.3, 359-368
- [Daw16] C. Daw, *The André-Oort conjecture via o-minimality* in O-minimality and Diophantine Geometry, edited by G.O. Jones and A.J. Wilkie, Cambridge University Press
- [DawOrr15] C. Daw, M. Orr, Heights of pre-special points of Shimura varieties, arXiv:1502.00822, to appear in Math. Annalen
- [De71] P. Deligne, Travaux de Shimura, Séminaire Bourbaki Exposé 389, LNM 244 (1971), 123-165
- [De79] P. Deligne, Variétés de Shimura: interprétation modulaire et techniques de construction de modèles canoniques, dans Automorphic Forms, Representations, and L-functions part. 2; Editeurs: A. Borel et W Casselman; Proc. of Symp. in Pure Math. 33, American Mathematical Society (1979) p. 247-290.
- [vdD98] L. van den Dries, Tame Topology and o-minimal structures. LMS lecture note series, 248, Cambridge University Press, 1998.
- [vdDM85] L. van den Dries, C. Miller On the real exponential field with restricted analytic functions, Israel J. Math. 85 (1994), 19–56.
- [Ed98] B. Edixhoven, Special points on the product of two modular curves, Compos. Math. 114, 315-328 (1998)
- [Ed05] B. Edixhoven, Special points on products of modular curves, Duke Math. J. 126, 325-348 (2005)
 [EdYa03] B. Edixhoven, A. Yafaev, Subvarieties of Shimura varieties, Annals of Mathematics (2) 157 (2003) 621-645
- [EMO] B. Edixhoven, B. Moonen, F. Oort (Editors) Open Problems in Arithmetic Geometry. Bull. Sci. Math. 125 (2001) 1-22
- [Ga68] A.M Gabrielov, Projections of semi-analytic sets, Funkt. Ana. i Prilozen 2 (1968), 18-30
- [Gao16] Z. Gao, Towards the André-Oort conjecture for mixed Shimura varieties: the Ax-Lindemann theorem and lower bounds for Galois orbits of special points, J. Reine. Angew. Math., to appear
- [Ge60] A.O. Gelfond, Transcendental and algebraic numbers, Dover Publications, New York (1960)
- [Gro84] A. Grothendieck, Esquisse d'un programme in Geometric Galois Actions vol. I, LMS Lecture Notes 242
- [GR14] E. Gaudron, G. Rémond, Polarisations et isogénies Duke Math. J. 163 (2014), 631-635
- [Hin88] M. Hindry, Autour d'une conjecture de Serge Lang, Invent. Math. 94, 575-603 (1988)
- [HwTo02] J.M. Hwang, W.K. To, Volumes of complex analytic subvarieties of Hermitian symmetric spaces, Amer. J. Math., 124 (2002), 1221-1246
- [KY14] B. Klingler, A. Yafaev , The André-Oort conjecture, Annals of Math. 180 (2014) 867-925
- [KUY16] B. Klingler, E. Ullmo, A. Yafaev, *The hyperbolic Ax-Lindemann-Weierstraβ conjecture*, Publications mathématiques de l'IHÉS **123**, Issue 1, 333-360 (2016)
- [Lang66] S.Lang, Introduction to transcendental numbers, Addison-Wesley Publishing, (1966)
- [1] F. Lindemann, Über die Zahl π , Math. Ann. **20** (1882), 213-225
- [Mar91] Margulis G.A, Discrete subgroups of semisimple Lie groups, Ergebnisse der Mathematik und ihrer Grengebiete 17, Springer-Verlag, (1991)
- [Ma76] D. Masser, Linear forms in algebraic points of Abelian functions III, Proc. London Math. Soc. 33 (1976), no.3, 549-564

- [MaWü95] D. Masser, G. Wüstholz, Factorization estimates for Abelian varieties Publ. Math. IHES 81 (1995) 5-24
- [Mi05] J. Milne, *Introduction to Shimura varieties*. In Harmonic Analysis, the Trace Formula and Shimura Varieties (James Arthur, Robert Kottwitz, Editors) AMS, 2005, (Lectures at the Summer School held at the Fields Institute, June 2 June 27, 2003). An expanded version is available for free download on author's web-page.
- [Mok89] N. Mok, Metric Rigidity Theorems on Hermitian Locally Symmetric Manifolds, Series in Pure Math. 6. World Scientific (1989).
- [Mok10] N. Mok, On the Zariski closure of a germ of totally geodesic complex submanifold on a subvariety of a complex hyperbolic space form of finite volume, Complex Analysis, 279-300, Trends Math., Birkhäuser Springer (2010).
- [Mok12] N. Mok, Extension of germs of holomorphic isometries up to normalizing constants with respect to the Bergman metric, J. Eur. Math. Soc. 14 (2012), 1617-1656
- $[\text{Moo98,I}] \ \ \text{B. Moonen}, \ \textit{Linearity properties of Shimura varieties.} \ \textit{I}, \ \text{J. Algebraic Geom.} \ \textbf{7} \ (1998), \ 539-567.$
- $[Moo98, II] \ B.\ Moonen,\ Linearity\ properties\ of\ Shimura\ varieties.\ II,\ Compositio\ Math.\ {\bf 114}\ (1998),\ 3-35.$
- [Mum79] D. Mumford, Hirzebruch's proportionality theorem in the non-cocompact case, Inventiones Math. 42, 239-272, 1979
- [Orr15] M. Orr, Families of abelian varieties with many isogeneous fibres, J. reine angew. Math. 705 (2015), 2117231
- [Oort94] F. Oort, Canonical liftings and dense sets of CM-points, Arithmetic Geometry (Cortona, 1994), 228-234, Symp. Math. XXXVII, Cambridge Univ. Press, Cambridge 1997
- [PS08] C. Peters and J. Steenbrink, Mixed Hodge Structures, volume 52 of A Series of Modern Surveys in Mathematics. Springer, 2008.
- [PetStar13] Y. Peterzil, S. Starchenko, Definability of restricted theta functions and families of abelian varieties, Duke Math. J. 162, (2013), 731-765
- [PetStar10] Y. Peterzil, S. Starchenko *Tame complex analysis and o-minimality*. Proceedings of the ICM, Hyderabad, 2010. Available on first author's web-page.
- [PW06] J. Pila, A. Wilkie, The rational points on a definable set, Duke Math. Journal 133, (2006) 591-616.
- [PiZa08] J. Pila, U. Zannier, Rational points in periodic analytic sets and the Manin-Mumford conjecture. Atti Accad. Naz. Lincei Cl. Sci. Fis. Mat. Natur. Rend. Lincei (9) Mat. Appl. 19 (2008), no. 2, 149-162.
- [Pil11] J. Pila, O-minimality and the Andre-Oort conjecture for \mathbb{C}^n , Annals Math. **173** (2011), 1779-1840. [PT14] J. Pila, J. Tsimerman, Ax-Lindemann for \mathcal{A}_g . Ann. of Math **179** (2014), 659-681
- [Pil] J. Pila, O-minimality and Diophantine geometry, Proceedings ICM 2014, to appear
- [Pink89] R. Pink, Compactification of mixed Shimura varieties, Dissertation, Bonner Math. Schriften
- [Pink05] R. Pink, A combination of the conjectures of Mordell-Lang and André-Oort, in Geometric Methods in Algebra and Number Theory, Progress in Math. 253 Birkhaüser (2005), 251-282
- [PS69] I.I. Pyateskii-Shapiro, Automorphic functions and the geometry of classical domains, translated from the Russian, Mathematics and its Applications, Vol. 8, Gordon and Breach Science Publishers (1969)
- [R] G. Rémond, Conjectures uniformes sur les variétés abéliennes, Preprint, 2015.
- [Ray88] M. Raynaud, Sous-variétés d'une variété abélienne et points de torsion, in Arithmetic and Geometry, Vol. I, Progress in Math. 35 (1988)
- [Sca12] T. Scanlon, A proof of the André-Oort conjecture via mathematical logic [after Pila, Wilkie and Zannier], Séminaire Bourbaki 1037, Astérisque no. 348 (2012), 299-315
- [Sca16] T. Scanlon, O-minimality as an approach to the André-Oort conjecture, preprint 2012 to appear in Panorama et Synthèse
- [ShWo95] H. Shiga, J. Wolfart, Criteria for complex multiplication and transcendence properties of automorphic functions, J. Reine Angew. Math. 463 (1995) 1-25
- [Schn37] T. Schneider, Arithmetische Untersuchungen elliptischer Integrale, Math. Annalen 113 (1937), no.1, 1-13

[Tsi12] J. Tsimerman, Brauer-Siegel for arithmetic tori and lower bounds for Galois orbits of special points, J. Amer. Math. Soc. 25 (2012), 1091-1117

[Tsi] J. Tsimerman, A proof of the André-Oort conjecture fo A_g , preprint 2015

[Panorama] Intersections exceptionnelles et conjectures de Zilber-Pink, volume of Panorama et Synthèse, to appear

[U07] E. Ullmo, Equidistribution des sous-variétés spéciales II, J. Reine Angew. Math. 606 (2007), 193-216

[U14] E. Ullmo, Applications du théorème d'Ax-Lindemann hyperbolique, Compositio Math. 150 (2014), 175-190

[U16] E. Ullmo, Structures spéciales et problème de Zilber-Pink, Introduction to the volume [Panorama]
 [UY14a] E. Ullmo, A. Yafaev, Galois orbits and equidistribution of special subvarieties: towards the André-Oort conjecture, Annals of Math. 180 (2014) 823-865

[UY11] E. Ullmo, A. Yafaev, A characterization of special subvarieties. Mathematika 57 (2011) 263-273
 [UY14b] E. Ullmo, A. Yafaev, Hyperbolic Ax-Lindemann theorem in the cocompact case, Duke Math. J. 163 (2014) 433-463

[UY15] E. Ullmo, A. Yafaev, Nombre de classes des tores de multiplication complexe et bornes inférieures pour orbites Galoisiennes de points spéciaux, Bull. Soc. Math. France 143 (2015), no.1, 197-228

[V] E. Villani, Mesures d'independance linéaire simultanées sur les périodes d'intégrales abéliennes.,
 Phd Thesis, University of Paris 6 (2005)

[Wil96] J. Wilkie, Model completeness results for expansions of the ordered field or real numbers by restricted Pfaffian functions and the exponential function, J. Amer. Math. Soc. 9 (1996), no.4, 1051-1094

[Wus87] G. Wüstholz, Algebraic groups, Hodge theory and transcendence, Proceedings of the International Congress of Mathematicians (Berkeley 1986), 476-483 (1987)

[Wus89] G. Wüstholz, Algebraische Punkte auf analytischen Untergruppen algebraischer Gruppen, Annals of Math. (2) 129 (1989), no.3, 501-517

[Ya05] A. Yafaev, On a result of Moonen on the moduli space of principally polarized Abelian varieties, Compos. Math 141, (2005), no. 5, 1103-1108

[Ya07] A. Yafaev, *The André-Oort conjecture-a survey*, in L-functions and Galois representations, 381-406, London Math. Soc. Lecture Note Ser. **320**, Cambridge University Press (2007)

[YuZh] X. Yuan, S. Zhang, On the averaged Colmez conjecture, preprint 2015

[Za12] U. Zannier, Some problems of unlikely intersections in arithmetic and geometry, with appendices by David Masser, Annals of Mathematics Studies 181, Princeton University Press (2012)

[Zil02] B. Zilber, Exponential sums equations and the Schanuel conjecture, J. London Math. Soc. (2) 65 (2002), no.1., 27-44

[Zil13] B. Zilber, Model theory of special subvarieties and Schanuel-types conjectures,

Bruno Klingler : Université Paris-Diderot (Institut de Mathématiques de Jussieu-PRG, Paris) and IUF.

email: bruno.klingler@imj-prg.fr.

Emmanuel Ullmo: Université Paris-Sud and IHES

email: ullmo@ihes.fr

Andrei Yafaev: University College London, Department of Mathematics.

email: yafaev@math.ucl.ac.uk